

Does the gender mix influence collective bargaining on gender equality? Evidence from France

Anne-Sophie Bruno, Nathalie Greenan, Jeremy Tanguy

▶ To cite this version:

Anne-Sophie Bruno, Nathalie Greenan, Jeremy Tanguy. Does the gender mix influence collective bargaining on gender equality? Evidence from France. 2021. halshs-03325842

HAL Id: halshs-03325842 https://shs.hal.science/halshs-03325842

Preprint submitted on 25 Aug 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



WORKING PAPER

N° 2021-7

DOES THE GENDER MIX INFLUENCE COLLECTIVE BARGAINING ON GENDER EQUALITY ? EVIDENCE FROM FRANCE

ANNE-SOPHIE BRUNO, NATHALIE GREENAN, JEREMY TANGUY

www.tepp.eu

TEPP – Theory and Evaluation of Public Policies - FR CNRS 2042

Does the gender mix influence collective bargaining on gender equality? Evidence from France

Anne-Sophie Bruno^{*} Nathalie Greenan[†] Jeremy Tanguy^{$\ddagger \$$}

Abstract

Gender equality at work has become in recent years a priority for governments. In France, collective bargaining is a main lever to achieve progress on gender equality issues. In a two-tier bargaining framework, industries and firms are required by law to negotiate on the reduction of gender inequalities. Using firm-level survey data on labor relations issues combined with administrative data, this paper seeks to better understand the dynamics of collective bargaining on gender equality at the firm level by questioning the role played by the gender mix. We find that gender diversity favors gender equality bargaining at the firm level. Under-representation and over-representation of women reduce the probability of firms negotiating an agreement on gender equality. The introduction of sanctions in the recent period has prompted low-feminized firms to negotiate more on gender equality but had little impact on highly feminized firms.

Keywords: gender equality, collective bargaining, gender diversity **JEL Classification**: J16, J52, C26

This paper has been accepted for publication in *Industrial Relations: A Journal of Economy and Society*

^{*}University Paris 1 - Panthéon Sorbonne, CHS. E-mail: Anne-Sophie.Bruno@univ-paris1.fr.

[†]CNAM, CEET, Lirsa, TEPP. E-mail: nathalie.greenan@lecnam.net.

[‡]Corresponding author. IREGE, University Savoie Mont Blanc. E-mail: jeremy.tanguy@univ-smb.fr. Address: IAE Savoie Mont Blanc, 4 chemin de Bellevue, BP 80439 - Annecy-le-Vieux, 74944 Annecy, France. Phone: +33 (0)4 50 09 24 53.

[§]We thank conference and seminar participants at TEPP 2018 conference, JMA 2019 conference (*Journées de Microéconomie Appliquée*) and University Savoie Mont Blanc for their helpful comments. We also thank Kadija Charni and Julien Cardoso for their research assistance. We acknowledge financial support from Agence nationale de la recherche (ANR) under the grant number ANR-16-CE26-0019-03. This work is also supported by a public grant overseen by ANR as part of the « Investissements d'avenir » program (reference : ANR-10-EQPX-17 – Centre d'accès sécurisé aux données – CASD). Any remaining errors are our own.

1 Introduction

Despite a great convergence between women and men in terms of education, labor market participation, occupations and working hours (Goldin 2014), women continue to earn lower wages in most developed countries (Blau and Kahn 2017). In 2019 in France, women's wages were 25% lower than men's all positions combined. For work of equal value, there remains a gap of 9% (French Ministry of Labor). Policy options to narrow both the gender gap in wages and in employment include more and more firm-level actions, such as adopting "equal pay for equal work" policies and the need to report firm-level pay gaps, or adopting regulations to increase the share of women in management and on boards (Magda and Cukrowska-Torzewska 2018). In several countries, collective bargaining is an important part of the "regulatory toolkit" to fight against gender inequalities in wages and working conditions (Milner et al. 2019). In France, firms are required to commit to the reduction of gender inequalities by negotiating collective agreements on these issues. While this obligation has existed since 2001, non-compliance is sanctioned only since 2012. Not all firms had complied with the law before the introduction of sanctions but also, to a lesser extent, after (Milner et al. 2019). There is therefore a stack in understanding what leads firms to negotiate a collective agreement on gender equality (GE thereafter).

In this paper we aim at evaluating the extent to which the presence of women inside the establishment contributes to the negotiation of collective agreements on GE. We investigate in particular the influence that the share of women in the workforce and among employee representatives have on the likelihood that negotiations on GE take place, and then result in a collective agreement or instead an action plan defined unilaterally by the employer. We first provide a descriptive analysis of differences in GE bargaining outcomes across French establishments depending on the share of women. We then assess the marginal effects on GE bargaining outcomes associated with different changes in the share of women throughout its distribution among French establishments.

Our empirical approach is based on the combination of establishment-, firm- and industry-level data. We mainly use the exhaustive data provided by the French Ministry of Labor on the collective bargaining texts (collective agreements and unilateral employer action plans), filed by establishments from a representative sample provided by a large-scale survey on labor relations issues for two recent periods (2008-2010, 2014-2016). To control for the endogenous nature of the share of women, we use an instrumental variable strategy based either on past variations or industry-level variations in the share of women, computed using administrative data. We take into account the role played by industry-level bargaining by comparing the bargaining behavior of establishments covered and not covered by an industry-level agreement on GE.

We find that contexts of gender diversity in the workforce and at the bargaining table are those that are the most favorable to collective bargaining on GE. The enforcement of sanctions for noncomplying establishments has sharply increased the propensity to negotiate in establishments with a more masculine workforce and employee representation, but has had little impact on the propensity to negotiate in establishments with a more feminized workforce and employee representation. When negotiations are undertaken, their outcome has clearly improved between the two periods. Over the most recent period, the probability that negotiations on GE lead to an agreement is close to 50%, whatever the gender composition of the workforce and of the negotiators. However, the chances that negotiations on GE result in an agreement are significantly lower in highly feminized bargaining environments.

We combine and contribute to three streams of literature.

We first contribute to the literature about collective bargaining on gender equality, which has already identified the presence of women in leadership and negotiating positions as one of the main factors providing the conditions for effective GE bargaining (Milner and Gregory 2014).¹ The main argument underlying this positive relationship between the presence of female negotiators and GE bargaining outcomes is that women give more importance to GE related issues. For instance, Heery and Kelly (1988) show that female British union officers are more likely to prioritize non-pay related issues -e.q. child care, maternity leave and sexual harassment - in collective bargaining than their male colleagues. Similarly, Harbridge and Thickett (2003) show in New Zealand that settlements covering predominately female populations of workers produced more advantageous family leave agreements than settlements covering predominately male populations, suggesting that female workers may be willing to trade off pay for other types of benefits. Although this role of women in helping effective GE bargaining has been shown by several authors (Dickens 1998, 2000; Gerstel and Clawson 2001; Williamson 2009, 2012), some others tend to downplay it (e.g. Heery 2006). Female representation and participation in union organizing would not be sufficient to make a difference, due to the domination of unions by men and masculine conceptions of work (Creese 1995; Danieli 2006; Jones 2002; Wajcman 2000). Male-dominated unions are seen as particularly uninterested in GE issues within the literature. They often associate the gender pay gap and work-life balance problems as "women's issues", which fit poorly with the masculine culture that characterizes male-dominated unions and workplaces (Gregory and Milner 2009; Kirton and Greene 2005). In addition, some authors qualify women's greater interest in these women's issues, such as Brochard and Letablier (2017) and Guillaume (2017) in France, who stress that equal pay and work-life balance receive little attention irrespective of the gender composition at the firm level. The diversity of findings in the literature can be explained by the scope analyzed in most existing studies on these issues, which are for the most part applied to a particular industry or a particular firm. In the present paper, we investigate the influence of the gender mix on GE bargaining outcomes for a representative sample of establishments in the French economy while controlling for a large set of establishment characteristics. This allows us to identify the marginal effect of increasing the share of women and to assess how this marginal effect varies depending on the initial balance of the gender mix.

The second stream of literature, that we relate to, focuses on gender differences around negotiation outcomes and propensity to negotiate. These gender differences in individual behavior are

^{1.} The other main factors identified in this literature include centralized bargaining systems, high bargaining coverage/union density, and supportive legislation.

related to collective bargaining with employer through the gender composition of the workforce and of representatives at the bargaining table. A well-known finding in this literature is that women are less likely to initiate negotiations ("women don't ask" in the words of Babcock and Laschever 2003), *i.e.* in the professional context to ask for promotions and pay raises. This finding is especially defended by laboratory experiments (e.g. Bowles et al. 2007, 2005; Dittrich et al. 2014) and field experiments (Leibbrandt and List 2015).² Some authors explain this finding by gender differences in the perception of bargaining. According to Babcock and Laschever (2003), the female gender role is incongruent with negotiating. In the same way, Leibbrandt and List (2015) show that women are less likely to negotiate because they perceive negotiating as a less acceptable behavior (than men). Gender differences in propensity to negotiate may also be explained by gender differences in implicit or explicit preferences, such as risk aversion or fairness concerns (see, e.q., Bertrand 2011; Croson and Gneezy 2009; Niederle 2016; Pfeifer and Stephan 2019).³ The extension of these findings outside the laboratory setting is nevertheless questioned by recent survey data studies (Artz et al. 2018; Säve-Söderbergh 2019; Stevens and Whelan 2019), which instead show that women are as likely as men to initiate negotiations.⁴ Once negotiations are initiated, most empirical evidence shows that women get lower outcomes than men (e.g. Artz et al. 2018; Castillo et al. 2013; Säve-Söderbergh 2019).⁵ This gender difference is partly explained by the experimental literature, which shows that women are less competitive, more cooperative and less assertive than men in bargaining situations (see Croson and Gneezy 2009). Beyond this average difference, women's bargaining performance depends on the context and the participants in the negotiation. Several authors indeed show that women negotiate better when negotiating with women (Bowles et al. 2007; Eckel and Grossman 2001; Hernandez-Arenaz and Iriberri 2018; Solnick 2007; Sutter et al. 2009). In addition, women negotiate better when they have a higher relative position power in the negotiation, *i.e.* when they act as employers and in matrilineal firms (Andersen et al. 2018; Dittrich et al. 2014).

Such heterogeneities in individual behaviors depending on the context of the negotiation suggest heterogenous effects of women presence on collective bargaining outcomes. We take into account this heterogeneity by controlling for a large set of establishment characteristics and considering different levels of women presence among employee representatives.

Last, our work relates to the recent literature analyzing the contribution of firms to the gender wage gap, as the result of different pay policies either within or between firms. Within-firm inequalities arise when women are less paid than comparably productive male coworkers while doing the same job in the same firm. This is called the bargaining effect because, beyond cases of discrimina-

^{2.} See also Azmat and Petrongolo (2014); Babcock and Laschever (2003), and surveys of Bertrand (2011); Croson and Gneezy (2009); Exley et al. (2020).

^{3.} Pfeifer and Stephan (2019) show that women perceive their wages more often as fair than men, while controlling for wages and working time, that may explain that they are less likely to negotiate than men.

^{4.} Stevens and Whelan (2019) show that women are not less likely to negotiate than men when they have the same opportunities to negotiate.

^{5.} In contrast, Stevens and Whelan (2019) find no significant gender difference in negotiation outcomes. Säve-Söderbergh (2019) provides evidence that women tend to ask for a lower wage than men. Artz et al. (2018) show that women are less likely than men to receive a wage increase, while they are equally likely to ask for a wage increase.

tion (Blau and Kahn 2017), such gender inequalities would result mainly from the fact that women do not bargain their wages as well as men do (see Bertrand 2011). Firms also contribute to the gender wage gap due to gender segregation and/or sorting across firms, *i.e.* women are less likely to work at high-paying firms than men (see, e.g., Barth et al. 2016; Bayard et al. 2003; Cardoso et al. 2016; Groshen 1991; Ludsteck 2014; Sorkin 2017). In their seminal paper, Card et al. (2016) show that firm-level productivity premiums explain 21% of the gender wage gap in Portugal, with 15-20 pp due to the sorting effect, and 1-6 pp due to the bargaining effect. Using the same methodology, Coudin et al. (2018) show that firms account only for 8% of the gender wage gap in France, with a dominant role of the sorting effect and a very small (and even negative) bargaining effect. Bruns (2019) and Gallen et al. (2019) find similar results in Germany and Denmark, respectively. Labor market institutions may explain this lower contribution of firms to the gender wage gap, in particular the higher minimum wage in France (Coudin et al. 2018) and collective bargaining (Bruns 2019).⁶ The small (or negative) bargaining effect in France implies that gender inequalities in wages and employment are relatively low with French firms. This is something to keep in mind when analyzing the incentive of French firms to negotiate goals and measures to reduce gender inequalities, even if they may concern other aspects than wages and employment.

The remainder of the paper is organized as follows. Section 2 provides some institutional background on GE collective bargaining in France. Section 3 describes the different data used in the paper. Section 4 provides some descriptive evidence on differences in GE collective bargaining across establishments depending on the share of women. We explain the econometric approach in Section 5 and present the main results in Section 6. Section 7 concludes. Additional results are relegated to an extended appendix.

2 Institutional background

In France, it is through the negotiation of firm-level agreements that the legislator seeks to engage firms to act for equal employment and equal pay. Collective bargaining on GE in firms has gradually developed since the implementation of the *Roudy* Act in 1983, which introduced the obligation for the employer to submit a written report to the works council on the comparative situation of general employment and training conditions of women and men, called *Rapport de situation compare* (RSC). This act also gave, for the first time, the opportunity for employers to bargain with union delegates measures to reduce gender inequalities within the firm. In 2001, the *Génisson* Act turned this opportunity to bargain into an obligation, specifically on the firm's objectives in the area on GE at work and on the actions to achieve them. In addition to this mandatory negotiation on GE, the Act of March 23, 2006 initiated the definition and implementation of measures to remove gender wage inequality in the mandatory annual negotiations on the wage policy.

The period from 2010 onwards has been marked by an acceleration of firm level collective bar-

^{6.} Bruns (2019) finds that collective agreements compress the gender wage gap within firms.



Figure 1 – Number of texts on GE bargained at the firm and industry level

gaining (see Figure 1). This has been strongly promoted by the public authorities through a set of laws imposing sanctions but also setting the various procedures and parameters of the negotiations. It then became common practice to refer to *bargained public policy* (Groux 2005) or *state managed bargaining* (Mias et al. 2016). The Act of November 9, 2010 introduced for the first time financial penalties (up to 1% of payroll) for firms with 50 employees and more not complying with the obligation to be covered by a collective agreement or, failing that, by a unilateral employer decision (action plan) for workplace GE. Negotiations on GE have then to be renewed after one year in case of an action plan, three years in case of a collective agreement. These sanctions were implemented from January 1st, 2012 and have since been reinforced by the prohibition to bid on public contracts. The Decree of December 18, 2012 made bargaining on the effective compensation area compulsory. Finally, the *Real equality* Act (2014) streamlined obligations into a unique global negotiation on gender professional and pay equality, the *Rebsamen* Act (2015) has consolidated and streamlined the obligations of negotiation, merging equality bargaining with quality of working life and the *El Khomri* Act (2016) has taken further disposition to develop collective bargaining and modernise social dialogue.

Milner et al. (2019) argue that in France, the "development of increasingly strong employer duties and complex compliance requirements has [...] created a distinctive model of 'bargained equality" (p. 277-278). Another feature of the French negotiated equality model is that it is based on a principle of complementarity, enshrined in law, between industry and firm level collective bargaining. Both levels have an obligation to negotiate and bargaining at the firm level is supposed to add to agreements set at the industry level (on pay scales, classifications, joint initiatives on

key issues of the industry) provisions tailored to address gender equality issues which are specific to the workplace. However, if legislation developments since 2013 have maintained pressure on GE bargaining, they have also generated uncertainties on enforcement mechanisms and procedures of agreements, especially between the time when the new laws were enacted and their decrees published. In particular, uncertainty touched the bargaining level (firm or industry) of the collective agreement required to be in compliance with the law and the validity period of the bargained texts. Furthermore, Mias et al. (2016) argue that by putting more pressure through financial penalties on the firm level, legal provisions have increasingly made the firm rather than the industry the locus of negotiation. This is likely to have driven the relationship between the two levels of negotiations from complementarity towards substitution, as illustrated by the fall in the number of industry level agreements between 2010 and 2016 while firm level collective bargaining accelerated.

Thus, despite a real impetus given to negotiations on GE at firm level, the effectiveness of the *bargained gender equality policy* has yet to be assessed. In this paper we will focus on two indicators: on the one hand, the compliance of firms with their legal obligations, and on the other hand, the signing of an agreement when the firms have actually negotiated.

3 Data

In this paper, we combine survey data on labor relations issues with administrative data for representative samples of French establishments.

The main data set consists in establishment- and firm-level informations collected in the RE-PONSE⁷ survey over its two most recent editions : 2011 and 2017. This survey is carried out by the French Ministry of Labor and covers several issues on labor relations and collective bargaining. In each edition of the survey, we mainly rely on data collected from the employer representative interviewed face to face in each establishment of the sample.⁸ Samples of the 2011 and 2017 surveys are composed of 4,023 and 4,363 establishments respectively, and are each representative of establishments of 10 employees and more from private and semi-public industries (excluding administration and agriculture). Most questions addressed to employer representatives cover the three-year period preceding the interview, *i.e.* 2008-2010 for the 2011 survey, and 2014-2016 for the 2017 survey. It is the case for the questions that we exploit to define our dependent variables, which indicate whether i) GE has been negotiated in the firm or the establishment; ii) the negotiation has resulted in a collective agreement at the firm- or establishment-level. The first question is slightly different between the two editions of the survey. While the 2017 survey questions refer to negotiations only, the 2011 survey questions refer to negotiations and discussions with respect to GE. Therefore, more firms are likely to be concerned in 2011 than in 2017, regardless of the actual differences in bargaining activity between the two editions. In addition, while the questions were addressed to all firms in

^{7.} REPONSE is the acronym of French RElations PrOfessionnelles et NégociationS d'Entreprises.

^{8.} In each edition of the REPONSE survey, data were also collected from an employee representative and a selection of employees in the establishment.

2011, they were addressed in 2017 only to firms with employee representatives (elected by employees or designated by labor unions).

Negotiations on GE have to be renewed every three year in case of collective agreement or over a shorter time period in other cases. Hence, we consider that firms with 50 employees and more that have not been negotiating on GE over each three-year period in the 2011 and 2017 surveys do not comply with their legal obligations. Furthermore, as we want to capture compliance with law using a similar question for 2011 and 2017, we focus our analysis on establishments from firms with 50 employees and more with at least an elected (or designated) employee representative. Our working samples are thus composed of 2,753 establishments in 2011 and 3,361 establishments in 2017.

As regards the outcomes of collective bargaining on GE, we make the choice to exploit exhaustive data on firm-level collective agreements and other texts related to collective bargaining, provided by the French Ministry of Labor over the period 2005-2016. By law, French firms are requested to report to the Ministry of Labor all collective agreements negotiated between employee and employer representatives. This also applies to other texts related to collective bargaining, such as disagreements, amendments or unilateral employer decisions (action plans in the field of GE). Information contained in these agreements is standardized by the Ministry of Labor to build a longitudinal firmlevel dataset : the D@ccord database. Beyond variables identifying the establishment or firm filing the agreement (name, city, identification number, industry, branch agreement identification), each collective agreement is characterized with the following variables: type of text (agreement, amendment, denunciation, disagreement, accession to an agreement, unilateral decision of the employer), signatory unit (establishment, firm, group, branch), topic(s) of the negotiation (including wages, bonuses, employment, hours, union rights, labor conditions, on-the-job training, GE), signatories (union delegate, mandated employee, elected employee representative, works council, employer only, single delegation), the unions present at the negotiation and those signatories of the agreement (CGT, CFTC, CFDT, CFE-CGC, FO, others). Unlike REPONSE data, this dataset is not subject to reporting bias on the bargaining outcomes and makes it possible to analyze other results than just signing a collective agreement on GE over the three-year period (e.q. filing an action plan). It also allows us to refine the information provided by survey data on the implementation of negotiations. Indeed, firms that claim having not negotiated on GE but that have filed an agreement on this topic have been reclassified as firms negotiating on GE.⁹

Beyond firm-level collective agreements, we also consider information on industry-level agreements.¹⁰ These agreements initially apply only to firms that belong to the employer associations engaged in the negotiations, but are generally extended to the whole industry in the months following the signing of the agreement. Information on industry-level agreements is public and available online. From this open data, we have retrieved information on agreements focused on GE. In partic-

^{9.} Over the period 2014-2016, about 65% of firms have negotiated on GE following reclassification while they were initially about 55%.

^{10.} Note that the industry refers here to a finer level (over 250 IDCC) in the classification of firms than the level we already control for in our estimates (21 sections). This finer level is called *branche*.

ular, we know when the agreement was signed, when it was extended to the whole industry (when appropriate) and until when it was or is valid. From the extension date of the agreement, every firm belonging to the corresponding industry can be considered as covered by that agreement. Thus, we are able to identify whether firms are covered or not by an industry-level agreement on GE.

The last data source is the Annual Declaration of Social Data (DADS ¹¹), a French administrative file coming from firms' social security records, *i.e.* the annual form that French firms have to fill for every employee subject to payroll taxes. As administrative data, DADS have the advantage to be exhaustive and so to cover all firms from the REPONSE survey. The French National Statistical Institute (INSEE) transforms the raw DADS data into several files available for researchers under restricted access. We specifically exploit the position files (*fichier postes*) to get further information on the structure of the workforce and wages at firm level. In particular, we recover the total share of women in the workforce within the establishment and within the firm (for multi-etablishment firms) using positions filled at December 31, each year. Then, to be consistent with the timetable of the REPONSE survey, we consider for each firm the average share of women over the two three-year periods (2008-2010, 2014-2016).

We are particularly interested in the role played by female negotiators on the employees' side at the bargaining table. Based on the REPONSE survey, we use the share of women among elected employee representatives as a proxy of the share of women among employee representatives at the bargaining table. This is a proxy because all elected employee representatives do not bargain with the employer in French firms. Until recently, only union delegates were authorized to do so. Among employee representatives, union delegates have the specificity to be designated by a labor union. Some of them are also elected as employee representatives. Previously and until 2008, an employee who was designated by one of the five major unions at the country level was automatically authorized to represent all employees in collective bargaining with the employer. Since 2008, to have the right to bargain with the employer, a union must have obtained at least 10% of votes at the first ballot of the professional elections, designed to elect the employee representatives. The REPONSE survey also provides information on several characteristics of the firm – or the establishment in case of multi-establishment firms –, which are deemed, according to the literature, to explain a significant part of differences across firms in the propensity to negotiate on GE : industry, workforce size¹², region¹³, presence of union representatives¹⁴, legal category¹⁵, publicly traded vs. privately held,

^{11.} Déclaration Annuelle de Données Sociales

^{12.} See details on these characteristics in Table A.1

^{13.} The historical 22 regions in mainland France are considered here, *i.e.* before the grouping into 13 regions operated by the territorial reform in January 2016. Controlling for the region is particularly relevant when examining bargaining outcomes on GE since certain regions have been the place of experiments designed to strengthen GE bargaining in SMEs, experiments conducted as part of the program *Territoires d'Excellence*.

^{14.} Several studies stress the importance of union representatives to get equal pay and work-life balance issues on the bargaining agenda (Heery 2006; Williamson 2012), although other studies mitigate their role given their weakness (Hantrais and Ackers 2005; Seeleib-Kaiser and Fleckenstein 2009) and lack of interest in these issues (Baird 2004).

^{15.} Five broad legal categories are distinguished: 1) simplified joint-stock companies, 2) other commercial companies, 3) legal person under public law subject to commercial law (public companies), 4) private law groupings (e.g. associations), 5) other organizations. In the literature, Figart et al. (2002) and Whitehouse et al. (2001) stress

belonging to a group vs. independent. If these characteristics are decisive in triggering negotiations on GE, we must control for them in order to correctly identify the effect of female presence on these bargaining outcomes.

4 Descriptive analysis

In this section, as in the rest of the paper, we focus on establishments from firms with 50 or more employees, smaller firms being not concerned with the obligation to negotiate on GE. The proportion of the latter firms having negotiated on GE over both periods is consequently very small (less than 20%). In addition, we focus on firms with employee representatives because, as mentioned earlier, information on GE bargaining is available only for these firms in the 2017 edition of the REPONSE survey.



Figure 2 – Proportion of firms negotiating on GE, with or without filing a text

Notes : Firms with 50 or more employees with employee representatives. Percentages were computed using the establishment weights provided in the REPONSE survey. The periods corresponding to the 2011 and 2017 surveys are 2008-2010 and 2014-2016, respectively. Over 2008-2010, only 0.07% firms have filed an action plan, a percentage too small to appear graphically in the left-hand side panel.

We first examine in Figure 2 the proportion of establishments having negotiated on GE and those having filed a text (agreement or action plan) on this topic over the two periods corresponding to the 2011 and 2017 surveys, *i.e.* 2008-2010 and 2014-2016. We compare at the same time the proportions obtained among establishments covered and those not covered by an industry-level agreement on GE. Overall, the proportion of establishments having negotiated on GE has slightly increased between the two periods, from 58% in 2011 to 65% in 2017. In addition, the proportion of establishments having filed a text following negotiations has greatly increased, from 10% in 2011 to

out a particularly important weight of GE bargaining in public organizations, that provide a more receptive context for union influence (Heery 2006).

more than 40% in 2017. These figures are in line with those provided by the French bureau of labor (*Direction Générale du Travail*), as mentioned by Pochic et al. (2019). Although the introduction of financial penalties has led to a decrease in the proportion of establishments not covered by a text despite negotiations, it has not eliminated this phenomenon, still representing about 30% in 2017. This high proportion may result from the assimilation by some employer representatives of mandatory annual negotiations as negotiations on GE.¹⁶ The introduction of financial penalties has led to closer propensities to negotiate between establishments covered and not covered by an industry-level agreement. However, the two types of establishments differ in 2017 in the outcome of negotiations: in establishments covered by an industry-level agreement, negotiations result less frequently in a local collective agreement than in an action plan (8.8% against 6.6%).

We then compare in Figure 3 the average share of women in the workforce of establishments depending on whether a negotiation on GE has taken place and resulted into a text. Are establishments involved in GE bargaining more feminized? Figure 3 rather describes the opposite. In both periods the average feminization rate is lower among establishments having negotiated on GE than among the others, but the difference is more pronounced in the more recent period due to the higher share of women in establishments without negotiation. However, there is hardly no difference in terms of gender mix between those establishments whose negotiations have resulted in a text (agreement or action plan) and those whose negotiations have not.

Figure 3 – Share of women in the workforce : no negotiation, negotiation with or without agreement



Notes : Establishments from firms with 50 or more employees and employee representatives. Percentages were computed using the establishment weights provided in the REPONSE survey.

Does it mean that the probability for a firm to negotiate and sign a collective agreement on GE depends on the gender mix of the workforce? We first propose a descriptive answer to this question

^{16.} Initially focused only on wages, the mandatory annual negotiation for firms has been extended in 2014 to other bargaining topics, including the gender pay gap.

in Figure 4. Based on the 2017 sample of establishments, we have first computed the deciles of the distribution of the share of women in the workforce, allowing us to divide each sample (2011, 2017) in 10 groups of establishments. Theoretically these groups have the same size in 2017. As there has been no significant changes in the distribution of the share of women between the two periods, they should also be balanced in 2011. Figure A.1 (Appendix A) shows that each group made up represents about 10% of the sample in 2017 but also in 2011, with small variations compared to 2017. We then compute for each group the proportion of establishments having negotiated on GE, and the proportion of these establishments for which negotiations on GE have resulted in an agreement and in an action plan over the corresponding three-year period. These proportions are reported in dashed line for 2011 and solid line for 2017. The left-hand side panel shows that the proportion of establishments with GE bargaining is lower in the tails than in the middle of the distribution, thus forming an inverted-U shape relationship with the share of women. In other words, establishments with a very low or a very high share of women are less likely to negotiate on GE than establishments where the gender mix is more balanced. This inverted-U shape is observed in both 2011 and 2017 but the proportions are much higher in 2017, except at the right end of the distribution where the proportions of the two periods are closer. We do not find the same shape when considering then the proportion of negotiating establishments having signed an agreement (see Figure 4, middle panel). While no clear relationship emerges in 2011, the probability of signing an agreement in 2017 is clearly higher in male-dominated and female-dominated establishments than in mixed establishments. However, as for the negotiation rate, the most feminized establishments (top 10%) report a signature rate which is lower with respect to other female-dominated establishments. In contrast, their likelihood of filing an action plan – used when negotiations could not lead to an agreement – is higher than that reported by other female-dominated establishments having negotiated on GE (see Figure 4, right-hand side panel). These highly feminized establishments thus seem to have a specific negotiation behavior that it will be worth investigating further in the upcoming econometric analysis. Note that both the signature rate for agreements and the filing rate for action plans have increased between 2011 and 2017 among negotiating establishments and this whatever the workforce gender mix. The signature rate hovers around 50% in 2017 while it is capped at 20% in 2011. The filing rate fluctuates between 10% and 20% in 2017 while it is almost uniformly zero in 2011, as action plans were not yet defined in law.

Beyond the share of women in the workforce, we then look at how GE bargaining outcomes vary depending on the share of female among employee representatives – thereafter share of female reps. As for the share of women, we describe these variations across different threshold values defined using the 2017 distribution of the share of female reps over all establishments with employee representatives belonging to firms with at least 50 employees. Unlike the share of women, defining 9 threshold values of the share of female reps – so as to have 10 same-size groups of establishments – is not relevant given the shape of the distribution (see Figure A.2 in Appendix A). Over both periods, we observe a large proportion of establishments where the share of female reps is zero or 100%. Each extreme case

Figure 4 – Proportion of establishments negotiating and then filing a text on GE by the share of women



Notes: Establishments from firms with 50 or more employees with employee representatives. Proportions were computed using the establishment weights provided in the REPONSE survey. The left-hand side panel covers all establishments while the two others cover only establishments negotiating on GE. The 10 groups reported on the x-axis are formed using the decile threshold values from the 2017 distribution of shares of women. The values of these thresholds as well as sample distributions across these thresholds are reported in Figure A.1 in Appendix A.

represents at each period about 20% of the sample. Between these two extreme cases, the density follows an inverted-U shape, so that establishments are concentrated around 50% of female reps. This particular distribution encourages us to consider apart these extreme cases and two relevant thresholds between them: 33% and 50%.¹⁷ Once groups of establishments are defined based on these threshold values, we then compute for each group the proportion of establishments having negotiated on GE, and then among the latter the proportion having signed an agreement and the proportion having filed an action plan. All these proportions are reported in Figure 5. Establishments where employee representatives are exclusively women are those where the negotiation rate is the lowest in both periods (see left-hand side panel). However, this is the only common feature between the two curves describing the negotiation rate of establishments by the share of female reps in 2011 and 2017. While the highest negotiation rate in 2011 is reached in establishments with female-dominated representatives (with at least one male representative), this is observed in 2017 in establishments where women are a minority among representatives (but still present). The negotiation rate has sharply increased between the two periods in the latter and particularly in establishments with less than one-third of female reps, where the negotiation rate has increased by almost 20 pp. In contrast,

^{17.} The extreme values are not exclusive to small establishments or firms, where the low number of employee representatives makes the shares 0% and 100% more likely. Indeed, the proportion of small firms is not especially high among those where employee representatives are exclusively women or exclusively men. Similarly, these extreme values are not exclusive to specific industries.

in establishments where women are the majority but not all representatives, the negotiation rate has hardly changed between the two periods. Among establishments having negotiated, the greater or lesser representation of women among representatives seems to have little impact on the likelihood of reaching an agreement or, conversely, of filing an action plan (see middle and right-hand side panels). Again, the most striking difference between 2011 and 2017 is the difference in level: almost 50% of establishments engaged in negotiations reached an agreement in 2017 while they were less than 20% in 2011.

Figure 5 - Proportion of establishments negotiating and then filing a text on GE by the share of female reps



Notes: Establishments from firms with at least 50 employees and employee representatives. Proportions were computed using the establishment weights provided in the REPONSE survey. The left-hand side panel covers all establishments while the two others cover only establishments having negotiating on GE over the previous three-year period. The 5 groups reported on the x-axis are formed using the threshold values reported in Figure A.3 in Appendix A.

At this stage, we cannot infer any impact of the gender mix on GE bargaining outcomes because establishments may differ in terms of other explanatory factors for bargaining depending on their share of women. This is why we propose in the next section a *ceteris paribus* analysis.

5 Identification strategies

We aim at assessing the causal relationship between the share of women in the establishment and the negotiation of a local collective agreement on GE, which can also lead to an action plan or no text at all if negotiations between the employer and employee representatives fail. We first investigate the effect of the share of women on the likelihood that negotiations on GE take place, starting from the following specification:

$$y_i^* = \mathbf{x}_i'\beta + \delta_1 \operatorname{Share}_i + \delta_2 \operatorname{Share}_i^2 + \varepsilon_i, \quad \varepsilon_i \sim N(0, 1), \quad y_i = 1(y_i^* > 0)$$
(1)

where y_i is a binary variable which is equal to 1 if negotiations on GE have taken place, 0 otherwise.¹⁸ y_i^* is the unobserved latent variable which is related to the observed y_i as described above. \mathbf{x}_i includes all relevant establishment characteristics used as control variables, as described in Section 3.¹⁹ Share_i refers to the share of women in the establishment *i*. share_i is also introduced in quadratic form to account for non-linearities in the effect of the share of women on the probability of negotiating on GE.²⁰ We assume that residuals ε_i are normally distributed, consistently with a probit specification.

In the second part of our analysis, we consider instead as dependent variable y_i an unordered categorical variable taking into account the different outcomes of GE bargaining : (1) no text, (2) agreement, and (3) action plan. To analyze this data, we use a multinomial response model, specifically a multinomial probit model, assuming the following latent threshold model:

$$y_{ij}^* = \mathbf{x}_i' \beta + \delta_1 \operatorname{Share}_i + \delta_2 \operatorname{Share}_i^2 + \varepsilon_i, \qquad (2)$$

where the ε 's follow a multivariate normal distribution and where the observable dependent variable y_i is linked with its latent counterpart y_i^* via:

$$y_{i} = \begin{cases} j \text{ if } y_{ij}^{*} = \max(y_{i1}^{*}, y_{i2}^{*}, y_{i3}^{*}) \\ 0 \text{ otherwise} \end{cases}$$
(3)

5.1 Endogenous share of women

Statistically, the coefficient estimated on Share_i in equations (1) and (2) may only report a correlation between that proportion and the outcome y_i , and not necessarily a causal relationship. Indeed, Share_i is likely to be correlated with the residual ε_i , due to plausible unobserved heterogeneity²¹ and reverse causation issues.

We propose two alternative strategies to improve identification of the causal relationship between share_i and the outcome. The first strategy consists in using the lagged value of the share of women, LagShare_i , instead of its current value among covariates. The model is then specified as follows :

$$y_i^* = \mathbf{x}_i' \beta + \delta_1 \operatorname{LagShare}_i + \delta_2 \operatorname{LagShare}_i^2 + \varepsilon_i, \tag{4}$$

As the share of women is computed as an average over the previous three-year period -2008-

^{18.} The reference period is the three-year period preceding the year of observation. Negotiations may have taken place both at establishment level and firm level in the case of multi-establishment firms.

^{19.} These characteristics include the size of the firm, the industry, the major occupational category (between (i) managers and professionals, (ii) intermediate occupations, (iii) clerical, sales and service workers, (iv) blue-collar workers), union presence (one or more union delegates in the firm), capital holders (publicly traded vs. privately held), legal category, belonging to a group vs. independent.

^{20.} We have also tried a specification including in addition the cubic form of share_i among covariates but decided not to keep it as it does not add explanatory power and leads to the same results as with this specification.

^{21.} Among unobserved factors of GE bargaining that may be correlated with the share of women, there is for instance the employer's sensitivity to GE issues: an employer who is very sensitive to these issues will presumably be more inclined to hire and promote women, but also to negotiate on GE.

2010 for 2011 and 2014-2016 for 2017 –, we consider an equivalent time period when computing the lagged value of this share. Thus, LagShare_i refers to the average share of women over 2005-2007 for 2011 and over 2011-2013 for 2017. We can use this strategy only for the share of women in the workforce, not for the share of female reps, because information on the latter is available only from the REPONSE survey and thus not available outside the periods covered by the survey.²² In contrast, information on the share of women is derived from the DADS and thus available every year. Including the lagged value of the share of women allows us to address potential biases from reverse causation but not from unobserved heterogeneity. In fact, some unobserved factors of GE bargaining are likely to be correlated with the past share of women. In addition, the effect of the past share of women on GE bargaining may not reflect perfectly the true effect of the current share of women on GE bargaining.

Alternatively, as a second strategy, we propose to use the lagged value of the share of women as an instrument of the current share of women in a two-stage residual inclusion (2SRI) estimation framework (see *e.g.* Terza et al. 2008). The 2SRI estimator is a nonlinear extension of the conventional instrumental variable (IV) method. Instead of replacing the endogenous variable by the first-stage predictor in the second stage, as in the conventional IV methods, the 2SRI method includes the first-stage residual as an additional regressor in the second stage. First proposed by Hausman (1978) in the linear context, this method proves to be consistent in the class of nonlinear models, for which the two-stage predictor substitution is inconsistent (see Terza et al. 2008). The first-stage equation specifies the share of women (Share_i) as a function of exogenous variables, including those introduced in the y_i^* equation (**x**) and a variable that affects just the share of women (LagShare_i) :

$$\text{Share}_i = \mathbf{x}'_i \,\gamma + \theta \,\text{LagShare}_i + \nu_i,\tag{5}$$

LagShare_i is correlated with Share_i but not with the residual ε_i , so it can be excluded from the y_i^* equation. Both properties must be satisfied to ensure consistent IV estimates. A nonzero correlation between the instrument and the residual ε_i , as well as a weak correlation between Share_{i,t-1} and Share_i, can induce an inconsistency in the IV estimates that exceeds the inconsistency of naive estimates.

The second-stage equation is then specified as:

$$y_i^* = \mathbf{x}_i' \beta + \delta_1 \operatorname{Share}_i + \delta_2 \operatorname{Share}_i^2 + \delta_3 \hat{\nu}_i + \varepsilon_i, \tag{6}$$

where $\hat{\nu}_i$ is the residual from the first-stage equation (5). Implementing a significance test on the coefficient δ_3 provides a simple and direct way to test the assumption that Share_i is exogenous with respect to y_i^* . We will reject this assumption if δ_3 is significantly different from 0. This strategy

^{22.} Using information from the previous wave of the survey is not relevant because the survey waves are widely spaced over time -i.e. 6 years - and only a small proportion of establishments are surveyed two editions in a row.

allows us to address biases from unobserved heterogeneity. As previously, this strategy is possible only for the share of women, not for the share of female reps, due to the availability of lagged values.

To test exogeneity of the share of female employee representatives, we use as instrumental variable the average share of female employee representatives among other establishments belonging to the same 2-digit industry, as defined using the 88 divisions of the French nomenclature of activities (NAF). We can reasonably expect the share of female employee representatives in a given establishment to be positively correlated with the average share of female employee representatives in the rest of the industry. We compute the latter share using for each period the total sample of establishments surveyed and applying the establishment weights provided to ensure the representativeness of the sample. We proceed in the same way to compute the average share of women among other establishments belonging to the same industry, that we will consider as an alternative instrumental variable for the share of women if the lagged share of women turns out to be a weak instrument or/and a non-valid instrument. For each specification, we will check the strength and the validity of each candidate instrument, and will choose the instrument(s) satisfying these hypotheses.

5.2 Endogenous selection of bargaining establishments

To analyze the effect the share of women has on the probability of signing an agreement or, failing that, of filing an action plan among establishments having started negotiations on GE, we face an additional methodological challenge: the non-random selection of the sub-sample of firms having negotiated on GE. To address biases from such endogenous selection, we use a non-parametric procedure designed to impose to this subsample of establishments the industry-size composition of the whole sample of establishments. We rely on the cell-by-cell approach suggested by Lemieux (2002), which is equivalent to the well-known reweighting method of DiNardo et al. (1996) while being more flexible. This consists first of dividing the data into a limited number J of industry-size cells, using the following dummy variables : we consider 18 industries, as defined in Table A.1, and within each industry we distinguish 4 size groups (50-99, 100-199, 200-499, 500 employees and more). For each cell j, at period t, we then estimate a reweighting factor Ψ_{jt} to be used in the computation of the counterfactual sample weight ω_{it}^c :

$$\omega_{jt}^c = \Psi_{jt}\,\omega_{jt},\tag{7}$$

with ω_{jt} the original sample weight of cell j at period t. The reweighting factor Ψ_{jt} is defined as :

$$\Psi_{jt} = \frac{\eta_{jt}^c}{\eta_{jt}},\tag{8}$$

where η_{jt} corresponds to the share of cell j in the sub-sample and η_{jt}^c is the share of the same cell j in the whole sample, at period t. That is, the numerator stands for the counterfactual sample share of cell j that we want to impose to be identical in the subsample. The resulting counterfactual

sample weights allow to control for unobserved factors related to the industry and the size of the firm, which may affect both the decision to negotiate on GE and the decision to sign a collective agreement on this issue, and thus induce a selection bias in estimates.²³

6 Results

We first analyze the extent to which a stronger female presence, both within the workforce and among employee representatives, affects the likelihood of starting negotiations on GE in the establishment, and then investigate how it can affect the likelihood that these negotiations result in a collective agreement or another bargaining outcome.

6.1 On the probability to start negotiations on GE

The influence of the gender mix

Tuble 1 The millence of the share of women on the presasing of hegethemis of the								
	2011	2011	2011	2011	2017	2017	2017	2017
	(1)	(2)	(3)	(4)	(1)	(2)	(3)	(4)
ShareOfWomen	0.106	2.317^{**}	_	2.064^{***}	-0.428	3.012^{***}	_	2.650^{***}
	(0.249)	(0.712)		(0.470)	(0.270)	(0.773)		(0.570)
$ShareOfWomen^2$	-	-2.300**	_	-2.456***	-	-3.618***	_	-3.861***
		(0.768)		(0.497)		(0.782)		(0.636)
LagShareOfWomen	_	_	2.129^{**}	_	_	_	2.611^{***}	—
-			(0.704)				(0.750)	
$LagShareOfWomen^2$	_	_	-2.312**	_	_	_	-3.110***	_
C C			(0.769)				(0.752)	
First-stage residual	_	_	_	0.571	_	_	_	0.807^{*}
				(0.335)				(0.403)
Observations	2,562	2,562	2,562	2,562	3,199	3,199	3,199	3,199

Table 1 - The influence of the share of women on the probability of negotiating on GE

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors and bootstrapped standard errors (200 replications) for the 2SRI estimator in column (4) are given in parentheses. All regressions include the set of controls defined in Section 5.

We first assess how the share of women affects the probability of negotiating on GE. Coefficient estimates for both periods are reported in Table 1. For each period, column (1) reports the coefficient on the share of women while assuming it is exogenous and has a linear effect on the probability of negotiating. Column (2) adds the quadratic form of the share of women, as in Equation 1, to possibly take into account its non-linear effect on the probability of negotiating. Column (3) replaces contemporaneous values by lagged values of the share of women and its quadratic form (see Equation 4), which are less or not affected by possible reverse causation coming from the dependent variable. Column (4) reports coefficient estimates from Equation 6, including the residual from the first-stage

^{23.} The conventional Heckman (1979) sample-selection correction procedure could not be used here in the absence of a relevant exclusion restriction that could be used in a first stage to explain the probability of negotiating on GE.

equation 5 to address potential biases from unobserved heterogeneity and reverse causation. ²⁴ The coefficient associated to this residual is non-significant in 2011, while it is weakly significant in 2017, implying that this specification is relevant only in 2017. Our preferred specification in 2011 is then that reported in column (2), where the significant coefficient on the quadratic term indicates that the effect of the share of women on the probability of negotiating is indeed nonlinear and so the specification reported in column (1) not relevant. Coefficient estimates from column (2) indicate that, in both periods, the share of women has a positive effect on the probability of negotiating but the magnitude of this effect decreases when the share of women increases until eventually it becomes zero or even negative. Such nonlinearity is consistent with the descriptive evidence reported in Figure 4 (left-hand side panel) and is persistent for both periods while controlling for endogenous issues in columns (3) and (4).



Figure 6 – The influence of the share of women on the probability of negotiating on GE

Notes : Estimates weighted using establishment weights provided in the REPONSE survey. Decile threshold values on the x-axis correspond to the following percentages : 1: 10.1%, 2: 18%, 3: 25.9%, 4: 37.2%, 5: 47.4%, 6: 56.5%, 7: 64%, 8: 72.8%, 9: 84.3%. "IV" estimates refer to estimates obtained using the 2SRI estimator. Probit estimates for 2017 are compared with 2017 IV estimates in Figure B.1 in Appendix B. IV estimates are preferred to probit estimates in 2017 given the non-zero coefficient on the first-stage residual in Table 1, column (8). All regressions include the set of controls defined in Section 5.

To better understand how this effect varies depending on the share of women in the establishment, we take again the 9 decile-thresholds of the 2017 distribution of the share of women (as defined in Figure A.1) and report in Figure 6 the probabilities of negotiating predicted using these threshold

^{24.} In both periods, we use as instrument the industry-level share of women, which is strongly correlated with the establishment-level share of women and not correlated with the dependent variable indicating negotiations on GE, as shown in Table B.1. We report at the top of this table the coefficient associated to each candidate instrument when included in the first-stage equation (see the left side) and in the second-stage equation (see the right side). In 2011, the industry-level share of women is preferred as instrument to the establishment-level lagged share of women because the latter is significantly correlated with the dependent variable. For comparison purposes, we use the same instrument in 2017 as in 2011, although the lagged share of women is apparently a valid instrument in 2017.

values (see left-hand side panel). Overall, controlling for differences in characteristics between establishments, we find again that GE bargaining has particularly increased between 2011 and 2017 in male-dominated establishments, more than in female-dominated establishments. Interestingly, the difference in probability between the two periods increases with the share of women: it is almost 20 pp at the 1st decile while it is less than 10 pp at the 9th decile. Otherwise, the probability of negotiating follows a similar inverted-U shape in 2011 and 2017 depending on the share of women. It increases with the share of women when they are poorly represented and decreases above a certain threshold when women are (highly) over-represented in the workforce, while it does not vary significantly between these two extreme cases. Due to the higher probability increase in the lower part of the distribution, the 2017 curve is flatter than the 2011 curve in this part of the distribution. In contrast, the 2017 curve is steeper than that of 2011 in the upper part of the distribution. In the right-hand side panel, we report changes in probability resulting from the transition from one decile-threshold to another and the corresponding 95% confidence intervals. The latter clearly indicate whether changes in probability are significantly different from 0 in the overall population of the establishments of interest, depending on whether they include or not zero. In 2011 transitions up to the 4th decile-threshold are each associated with an increase in the probability of negotiating - up to 5 pp -, while in 2017 only the transition from the 1st to the 2nd decile induces a significant increase in the probability of negotiating – with half the magnitude of 2011. Thus, increasing the share of women in male-dominated establishments was more decisive for GE bargaining in 2011 than in 2017. On the other side of the distribution, the probability of negotiating in 2017 starts to decline from a lower share of women and with greater magnitude than in 2011. 25

The literature provides different explanations for both the increase in the lower part and the decrease in the upper part of the distribution. These explanations are linked either to the overall share of women in the workforce or to the share of women among employee representatives. Indeed, the increased interest for women's issues at the bargaining table may be just linked to their increasing share in the workforce – *i.e.* the share of women in the workforce is now large enough for their working conditions to be discussed – or may also result from their greater influence at the bargaining table through their greater representation among employee representatives. The decline in GE bargaining above a certain share of women may come from a reduced interest for equal opportunities between men and women when the latter are largely over-represented in the workforce. This decline may also be linked to the lower propensity to negotiate of women (see Exley et al. 2020), who are likely to be the majority among employee representatives. To disentangle this specific role of female representatives from the overall influence of a greater female presence in the workforce, we then investigate the effect of the share of female representatives on the likelihood to initiate negotiations on GE.

^{25.} Transitions from the 6th decile-threshold are each associated to a decrease in the probability of negotiating in 2017, while only the transition from the 8th to the 9th decile-threshold is associated to a significant decrease in the probability of negotiating in 2011

The role of female representatives

	2011	2011	2011	2017	2017	2017
	(1)	(2)	(3)	(1)	(2)	(3)
ShareOfFemaleReps	-0.053	0.390	0.216	-0.182	1.035^{*}	1.156^{*}
	(0.131)	(0.396)	(0.444)	(0.146)	(0.433)	(0.461)
$ShareOfFemaleReps^2$	_	-0.451	-0.461	_	-1.222**	-1.220***
		(0.387)	(0.303)		(0.435)	(0.308)
First-stage residual	_	_	0.194	_	_	-0.129
			(0.349)			(0.364)
Observations	2,417	2,417	2,417	3,048	3,048	3,048

Table 2 - The influence of the share of female reps on the probability of negotiating on GE

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors and bootstrapped standard errors (200 replications) for the 2SRI estimator in column (3) are given in parentheses. All regressions include the set of controls defined in Section 5.

We report in Table 2 the coefficient estimates associated to the share of female representatives when explaining the probability of negotiating on GE. For each period, the specification in column (1) relies on the assumptions that the share of female representatives is exogenous and has a linear effect on the probability of negotiating. Column (2) adds the quadratic form of the share of female reps, as in Equation 1, to possibly take into account its non-linear effect on the probability of negotiating. Column (3) contains coefficient estimates from Equation 6, including the residual from the first-stage equation (5) to address potential biases from unobserved heterogeneity and reverse causation.²⁶ The coefficient associated to this residual is non-significant in both periods, implying that the specification in column (3) is not relevant. In 2017 the coefficient on the quadratic term is negative while the coefficient on the share of female reps is positive (see column (2)), suggesting that the share of female reps has a positive but decreasing effect on the probability of negotiating – as the share of women. In 2011 none of the coefficients reported in column (1) and (2) is significant, suggesting that the probability of negotiating on GE is not affected by the share of female reps over this period.

As for the share of women, we then compute the predicted probabilities at the different thresholds of the 2017 distribution of the share of female reps, as defined in Section 4, to get a better representation of its nonlinear effect (see Figure 7, left-hand side panel). The probability of negotiating follows an inverted-U shape depending on the share of female reps especially in 2017, the curve being rather flat in 2011. It is increasing between 0% and 33% of female reps, remaining constant up to 50% of female reps and then decreasing to reach its minimum for 100% of female reps. In the

^{26.} For each period, we use as instrument the industry-level share of female reps (without the establishment in question), which is strongly correlated with the establishment-level share of female reps but not correlated with the dependent variable indicating negotiations on GE, as shown in Table B.2. We report at the top of Table B.2 the coefficient associated to this instrumental variable when included in the first-stage equation (see the left side) and in the second-stage equation (see the right side).

Figure 7 – The influence of the share of female reps on the probability of negotiating on GE



Notes : Estimates weighted using establishment weights provided in the REPONSE survey. Threshold values on the x-axis correspond to the following percentages : 1: 0%, 2: 33%, 3: 50%, 4: 100%. All regressions include the set of controls defined in Section 5.

right-hand side panel of Figure 7, we report changes in probability resulting from the transition from one threshold to another as well as the corresponding 95% confidence intervals. The latter confirm the flat curve in 2011, as all include zero, while they support in 2017 the increasing probability in the first third and the decreasing probability in the second half of the distribution. While the 2017 curve mimics the curve obtained for the share of women over the same period, the 2011 curve does not replicate the increase in probability found on the first decile-thresholds of the share of women (see Figure 6). This suggests that the increased interest for women's issues at the bargaining table along the share of women is not specifically due to the influence of female reps. While existing literature has focused on the role of female negotiators in collective bargaining outcomes, we show that the weight of women in the workforce is already affecting the start of negotiations on GE.

Heterogeneity depending on the coverage by an industry-level agreement

The significant decrease in the probability of negotiating on GE found in 2017 in highly feminized establishments is at first sight surprising. With the introduction of sanctions against non-compliant establishments, we could expect a flatter curve in 2017, that would imply a lower dependence of GE bargaining on the gender composition. Instead, the decline in the probability of negotiating is further accentuated at the top of the distribution in 2017. In fact, the probability of negotiating has less increased in this part than in the rest of the distribution (see Figure 6 and Figure 7). This implies that a larger proportion of highly feminized establishments would have taken the risk of being financially sanctioned rather than negotiating an agreement. Given the legislative changes that have occurred each year during this period in the area of collective bargaining, we believe that

Figure 8 – Predicted probability of negotiating on GE depending on the coverage by an industrylevel agreement on GE



Notes : Predictions from regressions including the set of controls defined in Section 5 and weighted using establishment weights provided in the REPONSE survey. 95% confidence interval around each point estimate. All regressions include the set of controls defined in Section 5.

some establishments have not bargained on GE because of uncertainty as to the level of bargaining that would bring them into compliance with the law. Thus, rather than negotiating locally, several establishments may have chosen to apply an industry-level agreement – thereby believing to comply with the law. This behavior is likely to be more prominent in highly feminized establishments as they operate in industries where firm-level bargaining is traditionally less established.²⁷ In this section, we investigate whether the relatively low probability of negotiating on GE in highly feminized establishments may come from a greater use of industry-level agreements on this topic, by analyzing how the effect of the share of women varies depending on whether the establishment is covered by an industry-level agreement or not.

We first examine in Figure 8 whether establishments covered by an industry-level agreement on GE are less likely than other establishments to negotiate on this topic. In both periods, we find that the probability of negotiating on GE is not significantly different between establishments covered and those not-covered by an industry-level agreement. Although there is no difference on average between these two categories of establishments, we may still expect differences according to the share of women. Indeed, the probability of negotiating may differ between industry-level covered and uncovered establishments on specific deciles of the share of women. Figure 9 indicates that the proportion of establishments covered by an industry-level agreement differs strongly from a decile to another in both periods. The proportion of covered establishments decreases with the share of

^{27.} In particular, a large proportion of highly feminized establishments operate in the industry of Human health and social action, as shown in Figure A.6 in appendix.

Figure 9 – Distribution of deciles of the share of women depending on the coverage by an industrylevel agreement on GE



Note : Percentages were computed using establishment weights provided in the REPONSE survey.

women. In 2017, around 20% of the 10% most feminized establishments were covered by an industrylevel agreement, against almost 70% of the 10% least feminized establishments. However, while the proportion of covered establishments has increased in all deciles, it has particularly increased in the top deciles. Indeed, this proportion has increased by 124% in the highest decile of the share of women when the average increase is around 80%. Thus, despite the lower weight of industry-level agreements in the top deciles of the share of women, their particularly strong increase in this part of the distribution may contribute to explaining the sharper decline in the probability of negotiating in 2017.

We further examine whether the effect of the share of women differs according to the coverage of an industry-level agreement. To do this, we add in the previous regression an interaction term between the share of women and a dummy variable indicating the industry-level coverage. This interaction term allows us to estimate separately the effect of the share of women on the probability of negotiating according to whether or not establishments are covered by an industry-level agreement on GE. We report in Figure 10 the predicted probabilities of negotiating at the different deciles of the share of women among industry-level covered establishments (right-hand side panel) and uncovered establishments (left-hand side panel). The most striking fact is the larger decrease in 2017 in the probability of negotiating for high shares of women among industry-level covered establishments than among uncovered establishments. Unlike in 2011, in 2017 the decrease in this part of the distribution is significant whether or not an agreement exists at the industry-level, but it is much stronger in covered establishments. This is in line with our guess that highly feminized establishments have been more likely to apply an industry-level agreement on GE instead of negotiating a local agreement.



Figure 10 – The influence of the share of women on the probability of negotiating on GE depending on the coverage by an industry-level agreement on GE

Notes : Estimates weighted using establishment weights provided in the REPONSE survey. "IV" estimates refer to estimates obtained using the 2SRI estimator. IV estimates are preferred to probit estimates in 2017 given the non-zero coefficient on the first-stage residual in Table 1, column (8). All regressions include the set of controls defined in Section 5.

However, this specific behavior of highly feminized establishments does not explain totally their lower probability of negotiating in 2017, given the decrease also found in non-covered establishments. Therefore, this decrease can also be partly attributed to the gender composition of the workforce.

6.2 On the outcome of negotiations on GE

Once negotiations on GE have been initiated, do women play a role in the outcome of these negotiations? Do they facilitate the signing of an agreement on this issue? To answer that, we analyze in this section how variations in the total share of women in bargaining establishments affect the probability of reaching a text on gender equality, which can either be an agreement or an action plan. This distinction only makes sense in 2017, after the action plan has been legally defined as an alternative to the agreement for establishments to comply with the law. As 2011 precedes the law, texts on GE over the corresponding period include almost exclusively agreements.

The influence of the gender mix

Table 3 $-$ 1 he influence of the share of women on the probability of fling a text on GE									
	2011	2011	2011	2011	2017	2017	2017	2017	
	(1)	(2)	(3)	(4)	(1)	(2)	(3)	(4)	
ShareOfWomen	-1.195	-1.572	-1.303	-1.732^{*}	0.332	0.224	0.346	0.238	
	(0.989)	(0.989)	(0.680)	(0.681)	(0.810)	(0.831)	(0.539)	(0.539)	
$\mathrm{ShareOfWomen}^2$	1.098	1.568	1.137	1.625^{*}	-0.176	-0.156	0.022	-0.023	
	(1.057)	(1.060)	(0.732)	(0.732)	(0.854)	(0.861)	(0.606)	(0.606)	
First-stage residual	_	_	1.504	1.957	—	_	-1.906^{**}	-1.443^{*}	
			(1.110)	(1.114)			(0.715)	(0.708)	
Observations	1684	1684	1684	1684	2574	2574	2574	2574	
Weights	Standard	CF	Standard	CF	Standard	CF	Standard	CF	

. ...

. 1

. . . .

an

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors and bootstrapped standard errors (200 replications) for the 2SRI estimator in columns (3)-(4) are given in parentheses. All regressions include the set of controls defined in Section 5.

We first assess how the share of women affects the probability of filing a text on GE – agreements and action plans combined. Coefficient estimates for both periods are reported in Table 3. For each period, column (1) contains the coefficient on the share of women when assuming it is exogenous and allowing a non-linear effect on the probability of filing a text. Column (2) reports the same specification but using the counterfactual weights, instead of the standard weights provided in the survey, to control for selectivity into bargaining. Columns (3) and (4) add to columns (1) and (2), respectively, the residual from the first-stage equation to address potential biases from unobserved heterogeneity and reverse causation.²⁸ The coefficient associated to this residual is non-significant in

^{28.} For each period, we use as instrument the lagged share of women, which is strongly correlated with the share of women but not correlated with the dependent variable indicating a text on GE. We report in Table B.1 the coefficient associated to each candidate instrument when included in the first-stage equation (see the left side) and in the second-stage equation (see the right side), with standard weights (see middle part) and with counterfactual weights (see

2011, while it is weakly significant in 2017, implying that this specification is relevant only in 2017. Our preferred specification in 2011 is then that reported in column (2). Although the specification differs between the two periods, the result is the same: the share of women does not affect the probability of filing a text on GE, once negotiations on this issue have been engaged. This result is confirmed in Figure 11, where we report the following predicted probabilities of filing a text for the different deciles of the share of women (see left-hand side panel), changes in probability resulting from the transition from one threshold to another and the corresponding 95% confidence intervals. The latter clearly always include 0, indicating that none of the changes in probability is significantly different from 0. The outcome of negotiations on GE seems thus to be orthogonal to the share of women in the workforce in both periods.



Figure 11 – The influence of the share of women on the probability of filing a text on GE

Notes : Estimates weighted using establishment weights provided in the REPONSE survey. Decile threshold values on the x-axis correspond to the following percentages : 1: 10.1%, 2: 18%, 3: 25.9%, 4: 37.2%, 5: 47.4%, 6: 56.5%, 7: 64%, 8: 72.8%, 9: 84.3%. IV estimates refer to estimates obtained using the 2SRI estimator. Probit estimates for 2017 are compared with 2017 IV estimates in Figure B.2 in Appendix B. IV estimates are preferred to standard probit estimates given the non-zero coefficient on the first-stage residual in Table 3, column (8). CF means that counterfactual weights were applied to control for selection into the sample of establishments negotiating on GE. All regressions include the set of controls defined in Section 5.

This overall zero effect may hide in 2017 different effects of the share of women depending on the type of text considered. We then investigate how variations in the share of women affect the probability of signing an agreement on the one hand, and the probability of filing an action plan on the other hand. These two outcomes are analyzed jointly with respect to the no-text bargaining situation using the multinomial probit model described in Section 5. Coefficient estimates are reported in Table 4, in the upper part for the probability of signing an agreement and in the lower

bottom part). The lagged share of women is preferred as instrument to the industry-level share of women because the latter is significantly correlated with the dependent variable. We chose to select the same instrument in both periods, although the two candidates are valid in 2017, for comparison purposes between the two periods.

part for the probability of filing an action plan. Columns (1) and (2) contain the coefficient on the share of women when assuming it is exogenous, and applying the standard weights and counterfactual weights, respectively. Column (3) adds to column (1) the residual from the first-stage equation, to address potential biases from unobserved heterogeneity and reverse causation. The coefficient on the first-stage residual is weakly significant for both outcomes, implying that specification in column (3) is preferred to specification in column (1). However, the result does not differ between the two specifications: the share of women has no effect either on the probability of signing an agreement or on the probability of filing an action plan. The corresponding predicted probabilities (upper part) and changes in probability from one threshold to another (lower part) reported in Figure 12 confirm that all along the distribution of the share of women, there is no significant variation in the probability of signing an agreement or filing an action plan. Again, the outcome of negotiations on GE seems to be orthogonal to the share of women.

	(1)	(2)	(3)
	2017	2017	2017
Dependent variable:	Signing an	agreement	
ShareOfWomen	-0.460	-0.512	-0.443
	(1.143)	(1.185)	(0.773)
$ShareOfWomen^2$	1.006	0.829	1.261
	(1.216)	(1.246)	(0.840)
First-stage residual	_	_	-2.438^{*}
			(1.174)
Dependent variable:	Filing an a	ction plan	
ShareOfWomen	0.594	0.435	0.636
	(1.734)	(1.733)	(1.373)
$ShareOfWomen^2$	0.134	0.219	0.422
	(1.838)	(1.823)	(1.438)
First-stage residual	_	· _ /	-3.846*
-			(1.668)
Observations	2580	2574	2580
Weights	Standard	CF	Standard

Table 4 – The influence of the share of women on the probability of signing an agreement / filing action plan on GE (multinomial probit)

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors and bootstrapped standard errors (200 replications) for the 2SRI estimator (column 3) are given in parentheses. All regressions include the set of controls defined in Section 5.

While the share of women in the workforce plays a decisive role in initiating negotiations on GE, it seems not to affect the outcome of these negotiations. We investigate in the next section whether and to what extent the share of women among employee representatives plays a more decisive role in the outcome of these negotiations.



Figure 12 – The influence of the share of women on the probability of signing an agreement / filing an action plan on GE (multinomial probit)

Note : Estimates weighted using establishment weights provided in the REPONSE survey. Decile threshold values on the x-axis correspond to the following percentages : 1: 10.1%, 2: 18%, 3: 25.9%, 4: 37.2%, 5: 47.4%, 6: 56.5%, 7: 64%, 8: 72.8%, 9: 84.3%. All regressions include the set of controls defined in Section 5.

Role of the share of female employee representatives

	Table 6 The initialities of the share of female reps of the prosability of hing a text of of								
	2011	2011	2011	2011	2017	2017	2017	2017	
	(1)	(2)	(3)	(4)	(1)	(2)	(3)	(4)	
ShareOfFemaleReps	-0.661	-0.445	-1.352	-0.988	0.055	0.316	-0.343	-0.038	
	(0.526)	(0.546)	(0.764)	(0.542)	(0.484)	(0.487)	(0.654)	(0.433)	
$ShareOfFemaleReps^2$	0.622	0.325	0.521	0.243	0.003	-0.323	-0.019	-0.339	
	(0.523)	(0.541)	(0.517)	(0.433)	(0.466)	(0.470)	(0.470)	(0.326)	
First-stage residual	_	_	0.850	0.668	_	_	0.447	0.390	
			(0.600)	(0.460)			(0.542)	(0.352)	
Observations	1596	1596	1596	1596	2467	2467	2467	2467	
Weights	Standard	CF	Standard	CF	Standard	CF	Standard	CF	

Table 5 – The influence of the share of female reps on the probability of filing a text on GE

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors and bootstrapped standard errors (200 replications) for the 2SRI estimator in columns (3)-(4) are given in parentheses. All regressions include the set of controls defined in Section 5.

We follow with the share of female representatives the same steps as for the share of women in the workforce. Coefficient estimates for both periods are reported in Table 5, using the same specifications as in Table 3. The coefficient associated to the first-stage residual is non-significant in both periods ²⁹, implying that this specification is not relevant and the specification (2) will be preferred. Whatever the specification used and the period analyzed, the result is the same : the share of female reps does not affect the probability of filing a text on GE, once negotiations on this issue have been engaged. This zero effect is confirmed in Figure B.3 in Appendix B, where we show that changes in the probability of filing a text across the distribution of the share of female reps are not statistically significant.

Again, this zero effect can hide in 2017 different results for the two types of text considered. Using the same empirical strategy as the one followed for the share of women in the workforce, we then investigate how variations in the share of female reps affects the probability of signing an agreement on the one hand, and the probability of filing an action plan on the other hand. Coefficient estimates are reported in Table 6, in the upper part for the probability of signing an agreement and in the lower part for the probability of filing an action plan, using the same specifications as in Table 4 for the share of women. The coefficient on the first-stage residual is (weakly) significant only for the probability of signing an agreement. This implies that specification in column (3) will be preferred to specification in column (1) for this outcome only. As for the share of women, whatever the specification used, the coefficients suggest that each outcome is not affected by the share of female reps. However, the predicted probabilities reported in Figure 13 (upper part) give

^{29.} In both periods, we use as instrument the industry-level share of female reps (without the establishment in question), which is strongly correlated with the establishment-level share of female reps but not correlated with the dependent variable indicating a text on GE, as shown in Table B.2. We report in the middle part and the bottom part of Table B.2 the coefficient associated to the industry-level share of female reps when included in the first-stage equation (see the left side) and in the second-stage equation (see the right side), with standard weights and counterfactual weights, respectively.

a slightly different picture. Indeed, when using the IV specification – corresponding to column (3) in Table 6 –, the probability of signing an agreement declines when passing from one threshold to another (see left-hand side panel) while the probability of filing an action plan shows an increase (see right-hand side panel). The changes in probability and the corresponding 95% confidence intervals reported in the lower part of Figure 13 indicate that only the decrease in the probability of signing an agreement is significantly different from 0. Specifically, when the share of female reps goes from 50% to 100%, the probability of signing an agreement decreases by about 15 pp. There is also a significant decrease in that probability when the share of female reps goes from 33% to 50%, but it is smaller in magnitude (about 6 pp).

_	/		
	(1)	(2)	(3)
	2017	2017	2017
Dependent variable: S	ligning an ag	greement	
ShareOfFemaleReps	0.262	0.693	-0.663
	(0.683)	(0.692)	(0.556)
$ShareOfFemaleReps^2$	-0.121	-0.639	-0.175
	(0.664)	(0.672)	(0.443)
First-stage residual	_	_	1.044^{*}
			(0.444)
Dependent variable: H	Filing an act	ion plan	
ShareOfFemaleReps	-0.798	-0.904	-0.247
	(0.883)	(0.898)	(0.990)
$ShareOfFemaleReps^2$	1.053	1.059	1.090
	(0.867)	(0.885)	(0.624)
First-stage residual	_	· _ /	-0.634
-			(0.845)
Observations	2473	2467	2473
Weights	Standard	CF	Standard

Table 6 – The influence of the share of female reps on the probability of signing an agreement / filing action plan on GE (multinomial probit)

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors and bootstrapped standard errors (200 replications) for the 2SRI estimator (column 3) are given in parentheses. All regressions include the set of controls defined in Section 5.

To sum up, the place of women at the bargaining table has had in recent times an impact on the outcome of firm-level negotiations on GE, which was not the case over 2008-2010, before the recent reforms on GE bargaining. Surprisingly, more women at the bargaining table reduces the likelihood that negotiations on GE lead to an agreement. This does not necessarily imply that a strong female presence at the bargaining table undermines the success of negotiations on GE. Indeed, beyond signing an agreement, the success of negotiations on GE is also assessed by the quality of the agreement, *i.e.* the thickness of gender inequality reduction targets and actions defined to achieve them. The signing of an agreement could be less likely in very feminized bargaining environments, because the demands of the employee representatives for the reduction of gender inequalities are



Figure 13 – The influence of the share of female reps on the probability of signing an agreement / filing an action plan on GE (multinomial probit)

Notes : Estimates weighted using establishment weights provided in the REPONSE survey. Threshold values on the x-axis correspond to the following percentages : 1: 0%, 2: 33%, 3: 50%, 4: 100%. IV estimates refer to estimates obtained using the 2SRI estimator. CF means that counterfactual weights were applied to control for selection into the sample of establishments negotiating on GE (see section 5.2 for details). All regressions include the set of controls defined in Section 5.

more ambitious and less likely to get employer approval. Thus, agreements in these environments would be less frequent but more effective in addressing inequalities, an interesting hypothesis to test but which is outside the scope of this paper.

7 Conclusion

We analyze the influence of the gender mix in initiating collective bargaining on gender equality (GE) in French establishments and, once initiated, how it affects the outcome of these negotiations. Using administrative and survey data for a representative sample of French establishments over two recent periods (2008-2010 and 2014-2016), we show that contexts of gender diversity in the workforce are those that are the most favorable to initiate collective bargaining on GE in both periods. We indeed find that the probability of negotiating increases with the share of women when they are poorly represented and ends up decreasing when women become (strongly) over-represented in the workforce. This non-linear effect is an original result compared to existing evidence in the literature, which describes a single positive effect or no effect. In addition, while existing studies link the bargaining outcome to the gender of negotiators, we show that the gender composition of the workforce plays a role in initiating negotiations which is independent to that associated to the gender composition of employee representatives.

While there has been a general increase in GE bargaining between the two periods, due to financial penalties against firms not complying with the obligation to negotiate, the increase has been relatively more pronounced in male-dominated establishments and almost zero in highly feminized establishments. We show that the situation of the latter can be partly explained by the application of industry-level agreements instead of negotiating a local agreement, given temporary uncertainty on the right bargaining level to comply with law. Indeed, between 2014 and 2016, the law has modified each year the parameters of the French model of negotiated equality. This contributed to the disorientation of the actors and pushed some of them to postpone their negotiations.

When negotiations are undertaken, the likelihood of them leading to an agreement has increased overall with the introduction of financial penalties. However, the chances of reaching an agreement are significantly lower in highly feminized bargaining environments. Only the gender composition of employee representatives plays a role at this stage – not the composition of the workforce – which is consistent with the literature. However, we find a negative influence of the share of female representatives which does not echo results in existing studies.

This last finding leaves open the question of the role of women in the success of negotiations on GE. Indeed, it would be inappropriate to conclude that a greater female presence at the bargaining table undermines the success of negotiations on GE if it reduces the likelihood that an unambitious agreement will be signed in the fight against gender inequalities. Assessing the quality of collective agreements is particularly timely in the field of GE, where agreements are often suspected to be "empty shells". It is therefore a major challenge for the future to better understand the role of

women in GE bargaining.

References

- Andersen, S., Ertac, S., Gneezy, U., List, J. A., and Maximiano, S. (2018). On the Cultural Basis of Gender Differences in Negotiation. *Experimental Economics*, 21(4):757–778.
- Artz, B., Goodall, A. H., and Oswald, A. J. (2018). Do Women Ask? Industrial Relations: A Journal of Economy and Society, 57(4):611–636.
- Azmat, G. and Petrongolo, B. (2014). Gender and the Labor Market: What Have We Learned from Field and Lab Experiments? *Labour Economics*, 30(C):32–40.
- Babcock, L. and Laschever, S. (2003). Women Don't Ask : Negotiation and the Gender Divide. Princeton University Press, Princeton, NJ.
- Baird, M. (2004). Orientations to Paid Maternity Leave: Understanding the Australian Debate. The Journal of Industrial Relations, 46(3):259–275.
- Barth, E., Bryson, A., Davis, J. C., and Freeman, R. (2016). It's Where You Work: Increases in the Dispersion of Earnings across Establishments and Individuals in the United States. *Journal of Labor Economics*, 34(S2):S67–S97.
- Bayard, K., Hellerstein, J., Neumark, D., and Troske, K. (2003). New Evidence on Sex Segregation and Sex Differences in Wages from Matched Employee-Employer Data. *Journal of Labor Economics*, 21(4):887–922.
- Bertrand, M. (2011). New Perspectives on Gender. In Ashenfelter, O. and Card, D., editors, Handbook of Labor Economics, volume 4, chapter 17, pages 1543–1590. Elsevier.
- Blau, F. D. and Kahn, L. M. (2017). The Gender Wage Gap: Extent, Trends, and Explanations. Journal of Economic Literature, 55(3):789–865.
- Bowles, H. R., Babcock, L., and Lai, L. (2007). Social Incentives for Gender Differences in the Propensity to Initiate Negotiations: Sometimes It Does Hurt to Ask. Organizational Behavior and Human Decision Processes, 103(1):84 – 103.
- Bowles, H. R., Babcock, L., and McGinn, K. L. (2005). Constraints and Triggers: Situational Mechanisms of Gender in Negotiation. *Journal of Personality and Social Psychology*, 89(6):951–965.
- Brochard, D. and Letablier, M.-T. (2017). Trade Union Involvement in Work–Family Life Balance: Lessons from France. Work, Employment and Society, 31(4):657–674.
- Bruns, B. (2019). Changes in Workplace Heterogeneity and How They Widen the Gender Wage Gap. American Economic Journal: Applied Economics, 11(2):74–113.
- Card, D., Cardoso, A. R., and Kline, P. (2016). Bargaining, Sorting, and the Gender Wage Gap: Quantifying the Impact of Firms on the Relative Pay of Women. *The Quarterly Journal of Economics*, 131(2):633–686.
- Cardoso, A. R., Guimarães, P., and Portugal, P. (2016). What Drives the Gender Wage Gap? A Look at the Role of Firm and Job-Title Heterogeneity. Oxford Economic Papers, 68(2):506–524.
- Castillo, M., Petrie, R., Torero, M., and Vesterlund, L. (2013). Gender Differences in Bargaining Outcomes: A Field Experiment on Discrimination. *Journal of Public Economics*, 99:35 – 48.
- Coudin, E., Maillard, S., and To, M. (2018). Family, Firms and the Gender Wage Gap in France. IFS Working Papers W18/01, Institute for Fiscal Studies.

- Creese, G. (1995). Gender Equity or Masculine Privilege? Union Strategies and Economic Restructuring in a White Collar Union. The Canadian Journal of Sociology / Cahiers canadiens de sociologie, 20(2):143–166.
- Croson, R. and Gneezy, U. (2009). Gender Differences in Preferences. *Journal of Economic Literature*, 47(2):448–474.
- Danieli, A. (2006). Gender: The Missing Link in Industrial Relations Research. *Industrial Relations Journal*, 37(4):329–343.
- Dickens, L. (1998). Equal Opportunities and Collective Bargaining in Europe: Illuminating the Process. European Foundation for the Improvement of Living and Working Conditions, Dublin.
- Dickens, L. (2000). Collective Bargaining and the Promotion of Gender Equality at Work: Opportunities and Challenges for Trade Unions. *Transfer: European Review of Labour and Research*, 6(2):193–208.
- DiNardo, J., Fortin, N. M., and Lemieux, T. (1996). Labor Market Institutions and the Distribution of Wages, 1973-1992: A Semiparametric Approach. *Econometrica*, 64(5):1001–1044.
- Dittrich, M., Knabe, A., and Leipold, K. (2014). Gender Differences In Experimental Wage Negotiations. Economic Inquiry, 52(2):862–873.
- Eckel, C. C. and Grossman, P. J. (2001). Chivalry and Solidarity in Ultimatum Games. *Economic Inquiry*, 39(2):171–188.
- Exley, C. L., Niederle, M., and Vesterlund, L. (2020). Knowing When to Ask: The Cost of Leaning In. Journal of Political Economy, 128(3):816–854.
- Figart, D., Mutari, E., and Power, M. (2002). Living Wages, Equal Wages: Gender and Labour Market Policies in the United States. Routledge, London.
- Gallen, Y., Lesner, R. V., and Vejlin, R. (2019). The Labor Market Gender Gap in Denmark: Sorting Out the Past 30 Years. *Labour Economics*, 56:58–67.
- Gerstel, N. and Clawson, D. (2001). Unions' Responses to Family Concerns. Social Problems, 48(2):277–297.
- Goldin, C. (2014). A Grand Gender Convergence: Its Last Chapter. American Economic Review, 104(4):1091– 1119.
- Gregory, A. and Milner, S. (2009). Trade Unions and Work-life Balance: Changing Times in France and the UK? British Journal of Industrial Relations, 47(1):122–146.
- Groshen, E. L. (1991). The Structure of the Female/Male Wage Differential: Is It Who You Are, What You Do, or Where You Work? *The Journal of Human Resources*, 26(3):457–472.
- Groux, G. (2005). L'action publique négociée. Un nouveau mode de régulation ? Pour une sociologie politique de la négociation. *Négociations*, 3(1):57–70.
- Guillaume, C. (2017). Overcoming the Gender Pay Gap: Equal Pay Policies Implementation in France and the United Kingdom. In Auth, D., Hergenhan, J., and Holland-Cunz, B., editors, Gender and Family in European Economic Policy: Developments in the New Millennium, pages 63–80. Springer International Publishing, Cham.
- Hantrais, L. and Ackers, P. (2005). Women's Choices in Europe: Striking the Work-life Balance. European Journal of Industrial Relations, 11(2):197–212.
- Harbridge, R. and Thickett, G. (2003). Gender and Enterprise Bargaining in New Zealand: Revisiting the Equity Issue. New Zealand Journal of Employment Relations, 28(1):75.
- Hausman, J. A. (1978). Specification Tests in Econometrics. *Econometrica*, 46(6):1251–71.

Heckman, J. J. (1979). Sample Selection Bias as a Specification Error. *Econometrica*, 47(1):153–161.

- Heery, E. (2006). Equality Bargaining: Where, Who, Why? Gender, Work & Organization, 13(6):522–542.
- Heery, E. and Kelly, J. (1988). Do Female Representatives Make a Difference? Women Full-Time Officials and Trade Union Work. Work, Employment & Society, 2(4):487–505.
- Hernandez-Arenaz, I. and Iriberri, N. (2018). Women Ask for Less (Only from Men): Evidence from Bargaining in the Field. Journal of Economic Behavior & Organization, 152:192 – 214.
- Jones, S. (2002). A Woman's Place is on the Picket Line: Towards a Theory of Community Industrial Relations. *Employee Relations*, 24(2):151–166.
- Kirton, G. and Greene, A.-M. (2005). Gender, Equality and Industrial Relations in the 'New Europe': An Introduction. *European Journal of Industrial Relations*, 11(2):141–149.
- Leibbrandt, A. and List, J. A. (2015). Do Women Avoid Salary Negotiations? Evidence from a Large-Scale Natural Field Experiment. *Management Science*, 61(9):2016–2024.
- Lemieux, T. (2002). Decomposing Changes in Wage Distributions: A Unified Approach. Canadian Journal of Economics, 35:646–688.
- Ludsteck, J. (2014). The Impact of Segregation and Sorting on the Gender Wage Gap: Evidence from German Linked Longitudinal Employer-Employee Data. *ILR Review*, 67(2):362–394.
- Magda, I. and Cukrowska-Torzewska, E. (2018). Do Female Managers Help to Lower Within-Firm Gender Pay Gaps? Public Institutions vs. Private Enterprises. IZA Discussion Papers 12026, Institute of Labor Economics (IZA).
- Mias, A., Guillaume, C., Denis, J. M., and Bouffartigue, P. (2016). Vers un "dialogue social" administré? présentation du corpus. La nouvelle revue du travail, 8.
- Milner, S., Demilly, H., and Pochic, S. (2019). Bargained Equality: The Strengths and Weaknesses of Workplace Gender Equality Agreements and Plans in France. British Journal of Industrial Relations, 57(2):275–301.
- Milner, S. and Gregory, A. (2014). Gender Equality Bargaining in France and the UK: An Uphill Struggle? Journal of Industrial Relations, 56(2):246–263.
- Niederle, M. (2016). Gender. In Handbook of Experimental Economics, pages 481–563. Princeton University Press.
- Pfeifer, C. and Stephan, G. (2019). Why Women Do Not Ask: Gender Differences in Fairness Perceptions of Own Wages and Subsequent Wage Growth. *Cambridge Journal of Economics*, 43(2):295–310.
- Pochic, S., Brochard, D., Chappe, V., Charpenel, M., Demilly, H., Milner, S., and Rabier, M. (2019). L'égalité professionnelle est-elle négociable ? Enquête sur la qualité et la mise en oeuvre d'accords et de plans égalité élaborés en 2014 et 2015. Document d'études 231, DARES.
- Säve-Söderbergh, J. (2019). Gender Gaps in Salary Negotiations: Salary Requests and Starting Salaries in the Field. Journal of Economic Behavior & Organization, 161:35 – 51.
- Seeleib-Kaiser, M. and Fleckenstein, T. (2009). The Political Economy of Occupational Family Policies: Comparing Workplaces in Britain and Germany. British Journal of Industrial Relations, 47(4):741–764.
- Solnick, S. J. (2007). Gender Differences in the Ultimatum Game. *Economic Inquiry*, 39(2):189–200.
- Sorkin, I. (2017). The Role of Firms in Gender Earnings Inequality: Evidence from the United States. American Economic Review, 107(5):384–87.

- Stevens, K. and Whelan, S. (2019). Negotiating the Gender Wage Gap. Industrial Relations: A Journal of Economy and Society, 58(2):141–188.
- Sutter, M., Bosman, R., Kocher, M. G., and van Winden, F. (2009). Gender Pairing and Bargaining Beware the Same Sex! *Experimental Economics*, 12(3):318–331.
- Terza, J. V., Basu, A., and Rathouz, P. J. (2008). Two-Stage Residual Inclusion Estimation: Addressing Endogeneity in Health Econometric Modeling. *Journal of Health Economics*, 27(3):531–543.
- Wajcman, J. (2000). Feminism Facing Industrial Relations in Britain. British Journal of Industrial Relations, 38(2):183–201.
- Whitehouse, G., Zetlin, D., and Earnshaw, J. (2001). Prosecuting Pay Equity: Evolving Strategies in Britain and Australia. *Gender, Work & Organization*, 8(4):365–386.
- Williamson, S. (2009). Bargaining for Gender Equality in the Australian Public Service. Labour & Industry: a journal of the social and economic relations of work, 20(2):159–180.
- Williamson, S. (2012). Gendering the Bricks and Mortar: Building an Opportunity Structure for Equality Bargaining. Journal of Industrial Relations, 54(2):147–163.

A Appendix : Descriptive statistics

Figure A.1 – Sample distributions across decile thresholds of the share of women



B Appendix : Additional estimates



Figure A.2 – Kernel density estimate of the share of female reps

Figure A.3 – Sample distributions across thresholds of the share of female reps



Variables	REPONSE	REPONSE
	2011	2017
Total share of women	0.454	0.466
Share of female employee representatives	0.434	0.446
Presence of union representatives	0.780	0.747
Belonging to a group	0.578	0.645
Publicly traded	0.044	0.272
Legal category		
1. Simplified joint-stock companies	0.000	0.000
2. Other commercial companies	0.749	0.771
3. Public companies	0.035	0.014
4. Private law groupings (<i>e.g.</i> associations)	0.186	0.181
5. Other organizations	0.030	0.034
Firm sizo		
50 - 90 employees	0.226	0.204
100 - 100 employees	0.220	0.204 0.166
200 - 400 employees	0.170	0.100 0.146
500 employees and more	0.170	0.140 0.485
	0.111	0.400
Industry*		
B - Extractive industries	0.006	0.002
C - Manufacturing	0.142	0.125
D - Electricity, gaz, steam, aircon	0.011	0.008
E - Water production and distribution	0.013	0.015
F - Construction	0.050	0.059
G - Trading	0.219	0.213
H - Transport and storage	0.095	0.077
I - Accommodation and catering	0.047	0.047
J - Information and communication	0.028	0.038
K - Finance and insurance	0.054	0.046
L - Real estate activities	0.008	0.022
M - Specialized, scientific and technical activities	0.054	0.068
N - Administrative and support services	0.051	0.048
P - Teaching	0.029	0.027
Q - Human health and social action	0.156	0.182
R - Arts, entertainment and recreation	0.006	0.007
S - Other services	0.020	0.014
U - Extra-territorial activities	0.120	0.000
Observations	2,753	3,361

${\bf Table} ~ {\bf A.1} - {\rm Explanatory} ~ {\rm variables}: {\rm means} ~ {\rm and} ~ {\rm frequencies}$

Note : Figures are obtained using establishments weights provided in the REPONSE survey.

 * Sections from the French nomenclature of activities (NAF).



Figure A.4 – Firm-size distribution of the groups formed by share of women

Note : Establishments from firms with 50 or more employees with employee representatives. Percentages were computed using the firm weight variable provided in the REPONSE survey.



Figure A.5 – Proportion of establishments negotiating on gender equality by firm size

Notes: Establishments from firms with 50 or more employees with employee representatives. Percentages were computed using the firm weight variable provided in the REPONSE survey. The horizontal gray (black) dashed line indicates the average proportion of firms negotiating in 2011 (2017).



Figure A.6 – Industry distribution of the groups formed by share of women

Note : Establishments from firms with 50 or more employees with employee representatives. Percentages were computed using the firm weight variable provided in the REPONSE survey.

Industries : B - Extractive industries, C - Manufacturing, D - Electricity, gaz, steam, aircon, E - Water production and distribution, F - Construction, G - Trade, H - Transport and storage, I - Accommodation and catering, J - Information and communication, K - Finance and insurance, L - Real estate activities, M - Specialized, scientific and technical activities, N - Administrative and support services, P - Teaching, Q - Human health and social action, R - Arts, entertainment and recreation, S - Other services, U - Extra-territorial activities.



Figure A.7 – Proportion of establishments negotiating on gender equality by industry

Notes : Establishments from firms with 50 or more employees with employee representatives. Percentages were computed using the firm weight variable provided in the REPONSE survey. The horizontal gray (black) dashed line indicates the average proportion of firms negotiating in 2011 (2017).

Industries : B - Extractive industries, C - Manufacturing, D - Electricity, gaz, steam, aircon, E - Water production and distribution, F - Construction, G - Trade, H - Transport and storage, I - Accommodation and catering, J -Information and communication, K - Finance and insurance, L - Real estate activities, M - Specialized, scientific and technical activities, N - Administrative and support services, P - Teaching, Q - Human health and social action, R -Arts, entertainment and recreation, S - Other services, U - Extra-territorial activities.

	First stage				Second stage			
	2011	2011	2017	2017	2011	2011	2017	2017
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Dependent variable:		ShareO	fWomen			Negotiat	ingOnGE	
LagShareOfWomen	0.958***	_	0.917***	—	-2.722**	_	0.234	_
	(0.015)		(0.019)		(1.001)		(0.915)	
IndustryShareOfWomen	—	0.817^{***}	—	0.834^{***}	_	-0.467	—	-0.673
		(0.038)		(0.039)		(0.406)		(0.492)
Observations	2562	2562	3199	3199	2562	2562	3199	3199
Weights	Standard	Standard	Standard	Standard	Standard	Standard	Standard	Standard
Dependent variable:		ShareO	fWomen		FilingATextOnGE			
LagShareOfWomen	0.970^{***}	-	0.909^{***}	_	-1.459	_	1.724	_
	(0.008)	0.007***	(0.017)	0 700***	(1.352)	1 400**	(1.037)	0.050
IndustryShareOfWomen	_	0.807	_	0.796^{-10}	_	-1.420	_	0.850
	1004	(0.041)	0500	(0.040)	1004	(0.530)	0500	(0.460)
Observations	1684	1684	2580	2580	1684	1684	2580	2580
Weights	Standard	Standard	Standard	Standard	Standard	Standard	Standard	Standard
Dependent variable:		ShareO	fWomen			FilingAT	extOnGE	
LagShareOfWomen	0.967***	—	0.918***	—	-1.892	—	1.325	—
	(0.009)	0 0 0 * * *	(0.015)	0	(1.369)	1 005*	(1.099)	0.000
IndustryShareOfWomen	—	0.792***	—	0.772^{****}	—	-1.327*	—	0.602
		(0.042)		(0.045)		(0.521)		(0.476)
Observations	1684	1684	2574	2574	1684	1684	2574	2574
Weights	CF	CF	CF	CF	CF	CF	CF	CF

Table B.1 – Instrument strength and validity for the share of women

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors are given in parentheses. All regressions include the set of controls defined in Section 5.



Figure B.1 – The influence of the share of women on the probability of negotiating on GE

Notes : Estimates weighted using establishment weights provided in the REPONSE survey. Decile threshold values on the x-axis correspond to the following percentages : 1: 10.1%, 2: 18%, 3: 25.9%, 4: 37.2%, 5: 47.4%, 6: 56.5%, 7: 64%, 8: 72.8%, 9: 84.3%. "IV" estimates refer to estimates obtained using the 2SRI estimator. By default, estimates are obtained using a probit model. All regressions include the set of controls defined in Section 5.

	First	stage	Second stage		
	(1)	(2)	(3)	(4)	
	2011	2017	2011	2017	
Dependent variable:	ShareOfFe	emaleReps	Negotiati	ingOnGE	
IndustryShareOfFemaleReps	0.676^{***}	0.694^{***}	-0.483	0.075	
	(0.090)	(0.100)	(0.398)	(0.419)	
Observations	2417	3048	2417	3048	
Weights	Standard	Standard	Standard	Standard	
Dependent variable:	ShareOfFemaleReps		FilingATextOnGE		
IndustryShareOfFemaleReps	0.832^{***}	0.755^{***}	-0.789	-0.393	
	(0.107)	(0.111)	(0.533)	(0.440)	
Observations	1596	2473	1596	2473	
Weights	Standard	Standard	Standard	Standard	
Dependent variable:	ShareOfFe	emaleReps	FilingATe	extOnGE	
IndustryShareOfFemaleReps	0.814^{***}	0.698^{***}	-0.643	-0.268	
	(0.111)	(0.112)	(0.520)	(0.454)	
Observations	1596	2467	1596	2467	
Weights	CF	\mathbf{CF}	CF	CF	

 Table B.2 – Instrument strength and validity for the share of female reps

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Robust standard errors are given in parentheses. All regressions include the set of controls defined in Section 5.

Figure B.2 – The influence of the share of women on the probability of filing a text on GE in 2017



Notes : Estimates weighted using establishment weights provided in the REPONSE survey. Decile threshold values on the x-axis correspond to the following percentages : 1: 10.1%, 2: 18%, 3: 25.9%, 4: 37.2%, 5: 47.4%, 6: 56.5%, 7: 64%, 8: 72.8%, 9: 84.3%. "IV" estimates refer to estimate obtained using the 2SRI estimator. By default, estimates are obtained using a probit model. CF means that counterfactual weights were applied to control for selection into the sample of establishments negotiating on GE. All regressions include the set of controls defined in Section 5.



Figure B.3 – The influence of the share of female reps on the probability of filing a text on GE

Notes: Estimates weighted using establishment weights provided in the REPONSE survey. Threshold values on the x-axis correspond to the following percentages : 1: 0%, 2: 33%, 3: 50%, 4: 100%. CF means that counterfactual weights were applied to control for selection into the sample of establishments negotiating on GE (see section 5.2 for details). All regressions include the set of controls defined in Section 5.

21-6. The effects of the non-financial component of business accelerators Fabrice Gilles, Yannick L'Horty, Ferhat Mihoubi

21-5. Organisational changes and long term sickness absence and injury leave Mohamed Ali Ben Halima, Nathalie Greenan, Joseph Lanfranchi

21-4. The unexplored discriminations towards youth : equal access to goods and services David Gray, Yannick L'Horty, Souleymane Mbaye, Pascale Petit

21-3. The zero effect of income tax on the timing of birth: some evidence on French data Nicolas Moreau

21-2. Tropical cyclones and fertility : new evidence from Madagascar Idriss Fontaine, Sabine Garabedian, David Nortes-Martinez, Hélène Vérèmes

21-1. On the heterogeneous impacts of the COVID-19 lockdown on US unemployment Malak Kandoussi, François Langot

20-8. COVID-19 mortality and health expenditures across European countries: The positive correlation puzzle

Serge Blondel, Radu Vranceanu

20-7. Measuring discrimination in the labour market Emmanuel Duguet

20-6. The effects of age on educational performances at the end of primary school: crosssectional and regression discontinuity approach applications from Reunion Island Daniel Rakotomalala

20-5. Slowdown antitrust investigations by decentralization Emilie Dargaud, Armel Jacques

20-4. Is international tourism responsible for the pandemic of COVID19? A preliminary cross-country analysis with a special focus on small islands Jean-François Hoarau

20-3. Does labor income react more to income tax or means tested benefit reforms? Michaël Sicsic

20-2. Optimal sickness benefits in a principal-agent model Sébastien Ménard

20-1. The specific role of agriculture for economic vulnerability of small island spaces Stéphane Blancard, Maximin Bonnet, Jean-François Hoarau

19-8. The impact of benefit sanctions on equilibrium wage dispersion and job vacancies Sebastien Menard

19-7. Employment fluctuations, job polarization and non-standard work: Evidence from France and the US

Olivier Charlot, Idriss Fontaine, Thepthida Sopraseuth

19-6. Counterproductive hiring discrimination against women: Evidence from French correspondence test

Emmanuel Duguet, Loïc du Parquet, Yannick L'Horty, Pascale Petit

19-5. Inefficient couples: Non-minimization of the tax burden among French cohabiting couples

Olivier Bargain, Damien Echevin, Nicolas Moreau, Adrien Pacifico

19-4. Seeking for tipping point in the housing market: evidence from a field experiment Sylvain Chareyron, Samuel Gorohouna, Yannick L'Horty, Pascale Petit, Catherine Ris

19-3. Testing for redlining in the labor market Yannick L'Horty, Mathieu Bunel, Pascale Petit

19-2. Labour market flows: Accounting for the public sector Idriss Fontaine, Ismael Galvez-Iniesta, Pedro Gomes, Diego Vila-Martin

19-1. The interaction between labour force participation of older men and their wife: **lessons from France** Idriss Fontaine

18-15. Be healthy, be employed: a comparison between the US and France based on a general equilibrium model

Xavier Fairise, François Langot, Ze Zhong Shang

18-14. Immigrants' wage performance in the routine biased technological change era: France 1994-2012

Catherine Laffineur, Eva Moreno-Galbis, Jeremy Tanguy, Ahmed Tritah

18-13. Welfare cost of fluctuations when labor market search interacts with financial frictions

Elini Iliopulos, François Langot, Thepthida Sopraseuth

18-12. Accounting for labor gaps François Langot, Alessandra Pizzo

18-11. Unemployment fluctuations over the life cycle Jean-Olivier Hairault, Francois Langot, Thepthida Sopraseuth

18-10. Layoffs, Recalls and Experience Rating Julien Albertini, Xavier Fairise

18-9. Environmental policy and health in the presence of labor market imperfections Xavier Pautrel

18-8. Identity mistakes and the standard of proof Marie Obidzinski, Yves Oytana

18-7. Presumption of innocence and deterrence Marie Obidzinski, Yves Oytana

18-6. Ethnic Discrimination in Rental Housing Market: An Experiment in New Caledonia Mathieu Bunel, Samuel Gorohouna, Yannick L'Horty, Pascale Petit, Catherine Ris

18-5. Evaluating the impact of firm tax credits. Results from the French natural experiment CICE

Fabrice Gilles, Yannick L'Horty, Ferhat Mihoubi, Xi Yang

18-4. Impact of type 2 diabetes on health expenditure: an estimation based on individual administrative data

François-Olivier Baudot , Anne-Sophie Aguadé, Thomas Barnay, Christelle Gastaldi-Ménager, Anne Fargot-Campagna

18-3. How does labour market history influence the access to hiring interviews? Emmanuel Duguet, Rémi Le Gall, Yannick L'Horty, Pascale Petit

18-2. Occupational mobility and vocational training over the life cycle Anthony Terriau

18-1. Retired, at last? The short-term impact of retirement on health status in France Thomas Barnay, Eric Defebvre

17-11. Hiring discrimination against women: distinguishing taste based discrimination from statistical discrimination

Emmanuel Duguet, Loïc du Parquet, Pascale Petit

17-10. Pension reforms, older workers' employment and the role of job separation and finding rates in France

Sarah Le Duigou, Pierre-Jean Messe

17-9. Healthier when retiring earlier? Evidence from France Pierre-Jean Messe, François-Charles Wolff

17-8. Revisting Hopenhayn and Nicolini's optimal unemployment insurance with job search monitoring and sanctions

Sebastien Menard, Solenne Tanguy

17-7. Ethnic Gaps in Educational Attainment and Labor-Market Outcomes: Evidence from France

Gabin Langevin, David Masclet, Fabien Moizeau, Emmanuel Peterle

17-6. Identifying preference-based discrimination in rental market: a field experiment in Paris

Mathieu Bunel, Yannick L'Horty, Loïc du Parquet, Pascale Petit

17-5. Chosen or Imposed? The location strategies of households Emilie Arnoult, Florent Sari

17-4. Optimal income taxation with composition effects

Laurence Jacquet, Etienne Lehmann

17-3. Labor Market Effects of Urban Riots: an experimental assessment Emmanuel Duguet, David Gray, Yannick L'Horty, Loic du Parquet, Pascale Petit

17-2. Does practicing literacy skills improve academic performance in first-year university students? Results from a randomized experiment Estelle Bellity, Fabrices Gilles, Yannick L'Horty

17-1. Raising the take-up of social assistance benefits through a simple mailing: evidence from a French field experiment

Sylvain Chareyron, David Gray, Yannick L'Horty

16-8. Endogenous wage rigidities, human capital accumulation and growth Ahmed Tritah

16-7. Harder, better, faster...yet stronger? Working conditions and self-declaration of chronic diseases

Eric Defebvre

16-6. The influence of mental health on job retention Thomas Barnay, Eric Defebvre

16-5. The effects of breast cancer on individual labour market outcomes: an evaluation from an administrative panel

Thomas Barnay, Mohamed Ali Ben Halima, Emmanuel Duguet, Christine Le Clainche, Camille Regaert

16-4. Expectations, Loss Aversion, and Retirement Decisions in the Context of the 2009 Crisis in Europe

Nicolas Sirven, Thomas Barnay

16-3. How do product and labor market regulations affect aggregate employment, inequalities and job polarization? A general equilibrium approach

Julien Albertini, Jean-Olivier Hairault, François Langot, Thepthida Sopraseuth

16-2. Access to employment with age and gender: results of a controlled experiment Laetitia Challe, Florent Fremigacci, François Langot, Yannick L'Horty, Loïc Du Parquet, Pascale Petit

16-1. An evaluation of the 1987 French Disabled Workers Act: Better paying than hiring Thomas Barnay, Emmanuel Duguet, Christine Le Clainche, Yann Videau

15-10. Optimal Income Taxation with Unemployment and Wage Responses: A Sufficient Statistics Approach

Kory Kroft, Kavan Kucko, Etienne Lehmann, Johannes Schmieder

15-9. Search frictions and (in) efficient vocational training over the life-cycle Arnaud Chéron, Anthony Terriau

15-8. Absenteeism and productivity: the experience rating applied to employer contributions to health insurance

Sébastien Ménard, Coralia Quintero Rojas

15-7. Take up of social assistance benefits: the case of homeless Sylvain Chareyron

15-6. Spatial mismatch through local public employment agencies. Answers from a French quasi-experiment

Mathieu Bunel, Elisabeth Tovar

15-5. Transmission of vocational skills at the end of career: horizon effect and technological or organisational change

Nathalie Greenan, Pierre-Jean Messe

15-4. Protecting biodiversity by developing bio-jobs: A multi-branch analysis with an application on French data

Jean De Beir, Céline Emond, Yannick L'Horty, Laetitia Tuffery

15-3. Profit-Sharing and Wages: An Empirical Analysis Using French Data Between 2000 and 2007

Noélie Delahaie, Richard Duhautois

15-2. A meta-regression analysis on intergenerational transmission of education: publication bias and genuine empirical effect

Nicolas Fleury, Fabrice Gilles

15-1. Why are there so many long-term unemployed in Paris?

Yannick L'Horty, Florent Sari

14-14. Hiring discrimination based on national origin and the competition between employed and unemployed job seekers

Guillaume Pierné

14-13. Discrimination in Hiring: The curse of motorcycle women Loïc Du Parquet, Emmanuel Duguet, Yannick L'Horty, Pascale Petit

14-12. Residential discrimination and the ethnic origin: An experimental assessment in the Paris suburbs

Emmanuel Duguet, Yannick L'Horty, Pascale Petit

14-11. Discrimination based on place of residence and access to employment Mathieu Bunel, Yannick L'Horty, Pascale Petit

14-10. Rural Electrification and Household Labor Supply: Evidence from Nigeria Claire Salmon, Jeremy Tanguy

14-9. Effects of immigration in frictional labor markets: theory and empirical evidence from EU countries

Eva Moreno-Galbis, Ahmed Tritah

14-8. Health, Work and Working Conditions: A Review of the European Economic Literature Thomas Barnay

14-7. Labour mobility and the informal sector in Algeria: a cross-sectional comparison (2007-2012)

Philippe Adair, Youghourta Bellache

14-6. Does care to dependent elderly people living at home increase their mental health? Thomas Barnay, Sandrine Juin

14-5. The Effect of Non-Work Related Health Events on Career Outcomes: An Evaluation in the French Labor Market

Emmanuel Duguet, Christine le Clainche

14-4. Retirement intentions in the presence of technological change: Theory and evidence from France

Pierre-Jean Messe, Eva Moreno-Galbis, Francois-Charles Wolff

14-3. Why is Old Workers' Labor Market more Volatile? Unemployment Fluctuations over the Life-Cycle

Jean-Olivier Hairault, François Langot, Thepthida Sopraseuth

14-2. Participation, Recruitment Selection, and the Minimum Wage Frédéric Gavrel

14-1. Disparities in taking sick leave between sectors of activity in France: a longitudinal analysis of administrative data

Thomas Barnay, Sandrine Juin, Renaud Legal

13-9. An evaluation of the impact of industrial restructuring on individual human capital accumulation in France (1956-1993)

Nicolas Fleury, Fabrice Gilles

13-8. On the value of partial commitment for cooperative investment in buyer-supplier relationship

José de Sousa, Xavier Fairise

13-7. Search frictions, real wage rigidities and the optimal design of unemployment insurance Julien Albertini, Xavier Fairise

13-6. Tax me if you can! Optimal nonlinear income tax between competing governments Etienne Lehmann, Laurent Simula, Alain Trannoy

13-5. Beyond the labour income tax wedge: The unemployment-reducing effect of tax progressivity

Etienne Lehmann, Claudio Lucifora, Simone Moriconi, Bruno Van Der Linden

13-4. Discrimination based on place of residence and access to employment Mathieu Bunel, Emilia Ene Jones, Yannick L'Horty, Pascale Petit

13-3. The determinants of job access channels: evidence from the youth labor market in France Jihan Ghrairi

13-2. Capital mobility, search unemployment and labor market policies: The case of minimum wages

Frédéric Gavrel

13-1. Effort and monetary incentives in Nonprofit et For-Profit Organizations Joseph Lanfranchi, Mathieu Narcy

The TEPP Institute

The CNRS **Institute for Theory and Evaluation of Public Policies** (the TEPP Institute, FR n°2024 CNRS) gathers together research centres specializing in economics and sociology:

- L'Equipe de Recherche sur l'Utilisation des Données Individuelles en lien avec la Théorie Economique (Research Team on Use of Individuals Data in connection with economic theory), ERUDITE, University of Paris-Est Créteil and University of Gustave Eiffel
- Le Centre d'Etudes des Politiques Economiques de l'université d'Evry (Research Centre focused on the analysis of economic policy and its foundations and implications), EPEE, University of Evry Val d'Essonne
- Le Centre Pierre Naville (Research on Work and Urban Policies), CPN, University of Evry Val d'Essonne
- Le Groupe d'Analyse des Itinéraires et des Niveaux Salariaux (Group on Analysis of Wage Levels and Trajectories), GAINS, University of Le Mans
- Le Centre de Recherches en Economie et en Management, (Research centre in Economics and Management), CREM, University of Rennes 1 et University of Caen Basse-Normandie
- Le Groupe de Recherche ANgevin en Économie et Management (Angevin Research Group in Economics and Management), GRANEM, University of Angers
- Le Centre de Recherche en Economie et Droit (Research centre in Economics and Law)
 CRED, University of Paris II Panthéon-Assas
- Le Laboratoire d'Economie et de Management Nantes-Atlantique (Laboratory of Economics and Management of Nantes-Atlantique) LEMNA, University of Nantes
- Le Laboratoire interdisciplinaire d'étude du politique Hannah Arendt Paris Est, LIPHA-PE
- Le Centre d'Economie et de Management de l'Océan Indien, CEMOI, University of La Réunion

TEPP brings together 230 teacher-researchers and 100 doctoral students. It is both one of the main academic operators in the evaluation of public policies in France, and the largest multidisciplinary federation of research on work and employment. It responds to the demand for impact assessment of social programs using advanced technologies combining theoretical and econometric modeling, qualitative research techniques and controlled experiences.

www.tepp.eu